AS FILED WITH THE SECURITIES AND EXCHANGE COMMISSION ON JUNE 26, 1998
REGISTRATION NO. 333-

SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM S-8
REGISTRATION STATEMENT
UNDER
THE SECURITIES ACT OF 1933

FLOWSERVE CORPORATION

(Exact name of registrant as specified in its charter)

NEW YORK

31-0267900

(State or other jurisdiction of incorporation or organization)

(I.R.S. Employer Identification No.)

222 LAS COLINAS BOULEVARD, SUITE 1500

IRVING, TEXAS 75039
(Address of Principal Executive Offices)

- ------

THE DURIRON COMPANY, INC.,
PUMP AND FOUNDRY DIVISIONS HOURLY EMPLOYEES
SAVINGS AND THRIFT PLAN

- ------

RONALD F. SHUFF, ESQ.

VICE PRESIDENT, SECRETARY AND GENERAL COUNSEL FLOWSERVE CORPORATION

222 LAS COLINAS BOULEVARD, SUITE 1500

IRVING, TEXAS 75039

(972) 443-6500

(Name, address and telephone number, including area code, of agent for service)

CALCULATION OF REGISTRATION FEE

TITLE OF EACH CLASS OF SECURITIES TO BE REGISTERED	Amount to be registered	Proposed maximum offering price per security	Proposed maximum aggregate offering price	Amount of registration fee
Common Stock, par value \$1.25 per share(1)	50,000 shares	N/A	\$ 1,284,375(2)	\$378.89

- (1) This Registration Statement also covers the associated preferred stock purchase rights (the "Rights") issued pursuant to a Rights Agreement dated as of August 1, 1986, and amended as of August 1, 1996 and as of June 1, 1998, between the Registrant and National City Bank, as Rights Agent. Prior to the occurrence of certain events, the Rights will not be exercisable or evidenced separately from the Registrant's Common Stock.
- (2) Estimated solely for purposes of calculating the registration fee required by Section 6(b) of the Securities Act of 1933 (the "Securities Act"), the proposed maximum aggregate offering price of the Registrant's Common Stock

was calculated in accordance with Rule 457(c) and (h) under the Securities Act as the average of the high and low prices per share of the Registrant's Common Stock on June 23, 1998 as reported on the New York Stock Exchange, multiplied by 50,000, the number of shares of the Registrant's Common Stock to be offered or sold pursuant to the employee benefit plan described berein.

2

PART II

INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

ITEM 3. INCORPORATION OF DOCUMENTS BY REFERENCE

Flowserve Corporation (the "Registrant") hereby incorporates by reference into this Registration Statement on Form S-8 (the "Registration Statement") the following documents previously filed with the Securities and Exchange Commission (the "SEC") pursuant to the Securities Exchange Act of 1934, as amended (the "Exchange Act"):

- (a) The Registrant's Annual Report on Form 10-K for the fiscal year ended December 31, 1997;
- (b) The Registrant's Quarterly Report on Form 10-Q for the quarter ended March 31, 1998;
- (c) The description of the Registrant's Common Stock contained in the Registration Statement filed pursuant to Section 12 of the Exchange Act, including any amendment or report filed for the purpose of updating such description.

All documents subsequently filed by the Registrant pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date of this Registration Statement and prior to the filing of a post-effective amendment which indicates that all securities offered hereby have been sold or which deregisters all such securities then remaining unsold, shall be deemed to be incorporated by reference into this Registration Statement and to be a part hereof from the date of filing of such documents. Any statement contained in a document incorporated or deemed to be incorporated by reference herein shall be deemed to be modified or superseded for purposes of this Registration Statement to the extent that a statement contained herein or in any subsequently filed document which also is deemed to be incorporated by reference herein modifies or supersedes such statement. Any such statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part of this Registration Statement.

ITEM 4. DESCRIPTION OF SECURITIES

Not applicable.

ITEM 5. INTERESTS OF NAMED EXPERTS AND COUNSEL

None.

1

ITEM 6. INDEMNIFICATION OF DIRECTORS AND OFFICERS

The Business Corporation Law of the State of New York ("BCL") provides that if a derivative action is brought against a director or officer, the Registrant may indemnify him or her against amounts paid in settlement and reasonable expenses, including attorneys' fees incurred by him or her in connection with the defense or settlement of such action, if such director or officer acted on good faith for a purpose which he or she reasonably believed to be in the best interests of the Registrant, except that no indemnification shall be made without court approval in respect of a threatened action, or a pending action settled or otherwise disposed of, or in respect of any matter as to which such director or officer has been found liable to the Registrant. In a nonderivative action or threatened action, the BCL provides that the Registrant may indemnify a director or officer against judgments, fines, amounts paid in settlement and reasonable expenses, including attorneys' fees incurred by him or her in defending such action if such director or officer acted in good faith for a purpose which he or she reasonably believed to be in the best interests of the Registrant.

Under the BCL, a director or officer who is successful, either in a derivative or nonderivative action, is entitled to indemnification as outlined above. Under any other circumstances, such director or officer may be indemnified only if certain conditions specified in the BCL are met. The indemnification provisions of the BCL are not exclusive of any other rights to which a director or officer seeking indemnification may be entitled pursuant to the provisions of the certificate of incorporation or the by-laws of a corporation or, when authorized by such certificate of incorporation or by-laws, pursuant to a shareholders' resolution, a directors' resolution or an agreement providing for such indemnification.

The above is a general summary of certain indemnity provisions of the BCL and is subject, in all cases, to the specific and detailed provisions of Sections 721-725 of the BCL.

Article IX, Section 1 of the Registrant's By-laws provide that the Registrant shall indemnify any present or future director or officer from and against any and all liabilities and expenses to the maximum extent permitted by the BCL as the same presently exists or to the greater extent permitted by any amendment hereafter adopted.

Section 726 of the BCL also contains provisions authorizing the Registrant to obtain insurance on behalf of any such director and officer against liabilities, whether or not the Registrant would have the power to indemnify against such liabilities. As permitted by law, the Registrant maintains and pays premiums for directors' and officers' liability insurance policies.

ITEM 7. EXEMPTION FROM REGISTRATION CLAIMED

Not applicable.

2

4

ITEM 8. EXHIBITS

EXHIBIT NUMBER

- 4.1 Restated Certificate of Incorporation of the Registrant, as amended (filed as Exhibit 3.1 to the Registration Statement on Form S-4 as filed on June 19, 1997 (the "Form S-4")).*
- 4.2 By-Laws of the Registrant, as amended (filed as Exhibit 3.2 to the Form S-4).*
- 4.3 Rights Agreement dated as of August 1, 1986 between the Registrant and BankOne, N.A., as Rights Agent (filed as Exhibit 1 to the Registrant's Form 8-A dated August 13, 1986).*
- 4.4 Amendment dated as of August 1, 1996 to the Rights Agreement dated as of August 13, 1986 (filed as Exhibit 4.5 to the Registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 1996).*
- Amendment No. 2 dated as of June 1, 1998, to the Rights Agreement dated as of August 13, 1986, and amended as of August 1, 1996 (filed as Exhibit 1 to the Registrant's Form 8-A/A dated June 11, 1998).*
- 4.6 The Duriron Company, Inc., Pump and Foundry Divisions Hourly Employees Savings and Thrift Plan.
- 5.1 Opinion of Ronald F. Shuff.
- 23.1 Consent of Ronald F. Shuff (included in Exhibit 5.1).
- 23.2 Consent of Ernst & Young LLP
- 23.3 Consent of Price Waterhouse LLP

24.1 Powers of Attorney

* Incorporated by reference to a document previously filed with the SEC.

ITEM 9. UNDERTAKINGS

- (a) Rule 415 offering. The undersigned Registrant hereby undertakes:
- (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this Registration Statement:
- (i) To include any prospectus required by section 10(a)(3) of the Securities Act of 1933 (the "Act");

3

5

(ii) To reflect in the prospectus any facts or events arising after the effective date of the Registration Statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the Registration Statement;

(iii) To include any material information with respect to the plan of distribution not previously disclosed in the Registration Statement or any material change to such information in the Registration Statement.

provided, however, that paragraphs (a) (1) (i) and (a) (1) (ii) of this section do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in periodic reports

filed with or furnished to the Commission by the Registrant pursuant to section 13 or section $15\,(d)$ of the Exchange Act that are incorporated by reference in the Registration Statement.

- (2) That, for the purpose of determining any liability under the Act, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (b) Subsequent Exchange Act Documents. That, for purposes of determining any liability under the Act, each filing of the Registrant's annual report pursuant to section 13(a) or section 15(d) of the Exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to section 15(d) of the Exchange Act) that is incorporated by reference in the Registration Statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- Indemnification. Insofar as indemnification for liabilities (c) arising under the Act may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions, or otherwise, the Registrant has been advised that in the opinion of the Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

4

6

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, the Registrant certifies that it has reasonable grounds to believe that it meets all the requirements for filing this Form S-8 and has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Irving, State of Texas, on the 25th day of June 1998.

Flowserve Corporation (Registrant)

By: /s/ Ronald F. Shuff

Ronald F. Shuff Vice President, Secretary and General Counsel

Pursuant to the requirements of the Securities Act of 1933, this Registration Statement has been signed below by the following persons on June 10, 1998 in the capacities and on the date indicated.

Bernard G. Rethore	Chairman of the Board	June 25, 1	1998
/s/ WILLIAM M. JORDAN*			
William M. Jordan	Director	June 25, 1	L998
Hugh K. Coble	Director	June 25, 1	L998
/s/ DIANE C. HARRIS*			
Diane C. Harris	Director	June 25, 1	L998
/s/ GEORGE T. HAYMAKER*			
George T. Haymaker	Director	June 25, 1	L998
Michael F. Johnston	Director	June 25, 1	L998
/s/ CHARLES M. RAMPACEK*			
Charles M. Rampacek	Director	June 25, 1	1998
James O. Rollans	Director	June 25, 1	L998
/s/ WILLIAM C. RUSNACK*			
William C. Rusnack	Director	June 25, 1	1998

/s/ KEVIN E. SHEEHAN*			
Kevin E. Sheehan	Director	June 25,	1998
R. Elton White	Director	June 25,	1998
* By: /s/ RONALD F. SHUFF			
Ronald F. Shuff Attorney-in-fact			

EXHIBIT NUMBER	DESCRIPTION
4.1	Restated Certificate of Incorporation of the Registrant, as amended (filed as Exhibit 3.1 to the Registration Statement on Form S-4 as filed on June 19, 1997 (the "Form S-4")).*
4.2	By-Laws of the Registrant, as amended (filed as Exhibit 3.2 to the Form S-4).*
4.3	Rights Agreement dated as of August 1, 1986 between the Registrant and BankOne, N.A., as Rights Agent (filed as Exhibit 1 to the Registrant's Form 8-A dated August 13, 1986).*
4.4	Amendment dated as of August 1, 1996 to the Rights Agreement dated as of August 13, 1986 (filed as Exhibit 4.5 to the Registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 1996).*
4.5	Amendment No. 2 dated as of June 1, 1998, to the Rights Agreement dated as of August 13, 1986, and amended as of August 1, 1996 (filed as Exhibit 1 to the Registrant's Form 8-A/A dated June 11, 1998).*
4.6	The Duriron Company, Inc., Pump and Foundry Divisions Hourly Employees Savings and Thrift Plan.
5.1	Opinion of Ronald F. Shuff.
23.1	Consent of Ronald F. Shuff (included in Exhibit 5.1).
23.2	Consent of Ernst & Young LLP
23.3	Consent of Price Waterhouse LLP
24.1	Powers of Attorney

- -----

 $^{^{\}star}$ $\,$ Incorporated by reference to a document previously filed with the SEC.

EXHIBIT 4.6

CONFORMED COPY INCLUDES 1ST, 2D AND 3D AMENDMENTS

THE DURIRON COMPANY, INC., PUMP & FOUNDRY DIVISIONS HOURLY EMPLOYEES SAVINGS & THRIFT PLAN

THIS SAVINGS AND THRIFT PLAN AND TRUST AGREEMENT, by and among THE DURIRON COMPANY, INC., a corporation organized under the laws of the State of New York (hereinafter referred to as the "Company"), THE UNITED STEELWORKERS OF AMERICA, AFL-CIO-CLC, DISTRICT 30, LOCAL NUMBER 3320 (hereinafter referred to as the "Union") and the TRUSTEE;

WITNESSETH:

WHEREAS the Company entered into a collective bargaining agreement with the Union for its Employees who are members of the UNITED STEELWORKERS OF AMERICA, AFL-CIO-CLC, DISTRICT 30 LOCAL NUMBER 3320, under which retirement benefits were negotiated; and

WHEREAS to attain that end, the Company shall provide those Union Employees with a convenient way to save on a regular long-term basis as provided for in the collective bargaining agreement and, accordingly, has formulated the Plan herein embodied to comply with such terms; and

WHEREAS the form of this Agreement was established pursuant to the provisions of the Employee Retirement Income Security Act of 1974 ("ERISA"), the Tax Equity and Fiscal Responsibility Act of 1982 ("TEFRA"), the Retirement Equity Act of 1984, the Tax Reform Act of 1986 ("TRA"), the Omnibus Budget Reconciliation Acts of 1986 and 1987, the Technical and Miscellaneous Revenue Act of 1988, and the Omnibus Budget Reconciliation Act of 1989 and was approved by the Union and the Board of Directors of the Company.

NOW, THEREFORE, in consideration of the premises and of the mutual covenants contained herein, it is agreed by and between the Company, the Union, and the Trustee as follows:

2

CONFORMED COPY

ARTICLE I NAME

The Savings and Thrift Plan set forth herein shall be known as THE DURIRON COMPANY, INC., PUMP & FOUNDRY DIVISIONS HOURLY EMPLOYEES SAVINGS AND THRIFT PLAN and the Trust created in accordance with the terms hereof shall be known as THE DURIRON COMPANY, INC., PUMP & FOUNDRY DIVISIONS HOURLY EMPLOYEES SAVINGS AND THRIFT TRUST.

ARTICLE II DEFINITIONS

In this Plan, all nouns and pronouns used shall be construed in the singular or plural, and in such gender, as the context requires, and the following words and phrases shall be defined as set forth in this Article.

- 2.1 "Account Balance" shall mean the aggregate balance of the Employer Contribution Account, Employee Salary Deferral Account, Employee Matched Voluntary Contribution Account, Employee Nonmatched Voluntary Contribution Account, and Employee Rollover Account, as of a Determination Date.
- 2.2 "Actual Contribution Percentage" shall mean for each Plan Year the percentage derived by calculating the ratios (calculated separately for each Eligible Employee in the group) of: (a) the amount of any Company matching

contributions (Matching Company Contributions and Additional Company Matching Contributions) which are based upon an Employee's Salary Deferral or Matched Voluntary Contributions plus any Nonmatched Voluntary Contributions paid to the Trust Fund on behalf of each Eligible Employee for such Plan Year; to (b) the Eligible Employee's Compensation for such Plan Year. If so elected by the Plan Administrator, the numerator of the actual contribution percentage maybe calculated using Employee Salary Deferral Contributions and qualified nonelective contributions provided the conditions of Regulation Section $1.401\,(\mathrm{m})$ $-1\,(\mathrm{b})$ (2) are met. This percentage shall only be calculated if the Company provides for matching company contributions that would be classified as Company matching contributions as defined by Section 401(m) (4) of the Internal Revenue Code and those matching contributions meeting the requirements of Regulation Section 1.401(m)-1(b)(2) which are incorporated herein by reference and/or permit nondeductible Employee contributions to the Plan. As required by Section 401(m) (2) (B) of the Internal Revenue Code, for purposes of arriving at this percentage, the actual contribution ratio of a Highly Compensated Employee will be determined by treating all plans subject to Section 401(m) of the Internal Revenue Code and in which the Highly Compensated Employee is eligible, if required to be aggregated, as one plan.

- 2.3 "Actual Deferral Percentage" shall be calculated during each Plan Year and shall mean the average of the ratios (calculated separately for each Eligible Employee in the group) of:
- (a) The amount of Salary Deferral Contributions, additional Company contributions made pursuant to Section 4.8 and any other contributions that would fall within the confines of Section 401(k) (3) (D) of the Internal Revenue Code and satisfy the discrimination requirements as set forth in Regulation Section 1.401(k) -1(b) (3), the provisions of which are incorporated herein by reference, to be paid over to the Trust Fund on behalf of each such Eligible Employee for such Plan Year; to
- (b) The Eligible Employee's Compensation for such Plan Year.
- 2.4 "Adjustment Factor" shall mean the cost of living adjustment factor prescribed by the Secretary of the Treasury under Section 415(d) of the Internal Revenue Code for years beginning after December 31, 1987, as applied to such items and in such a manner as the Secretary shall provide. Such adjustment shall be effective as of the January 1st in the calendar year such increase is promulgated and applicable to the Plan Year which begins with or within such calendar year.
 - 2.5 "Anniversary Date" shall mean the 31st day of December each year.

-2-

3

CONFORMED COPY

[FIRST AMENDMENT REVISED SECTION TO READ AS FOLLOWS, EFFECTIVE 1/1/91]

- 2.6 "Annual Additions" shall mean the sum of Company contributions, Employee Salary Deferral Contributions (including excess contributions as defined in Section 401(k) (8) (B) of the Internal Revenue Code), nondeductible Employee Matched and Nonmatched Voluntary Contributions, the Participant's pro rata share of forfeitures, and amounts described in Sections 415(1) (2), 419(A) (d) (3), and 419(e) of the Internal Revenue Code, allocated to a Participant's account during the Plan Year.
- 2.7 "Bargaining Agreement" shall mean the labor agreement or agreements in force and effective between the Company and the UNITED STEELWORKERS OF AMERICA, AFL-CIO-CLC, DISTRICT 30 LOCAL NUMBER 3320, which covers rates of pay, hours of work, and other basic terms and conditions of employment.
 - 2.8 "Board of Directors" shall mean the Board of Directors of The

Duriron Company, Inc.

- 2.9 "Break in Service" shall mean a period of time which shall begin on an Employee's Termination Date and ends on the date the Employee returns to service and once again performs an Hour of Service for the Company, subject to the following.
- (a) Absence from employment at any time during a Plan Year by reason of service in the armed forces of the United States or, with the prior approval of the Company, by reason of service in a governmental agency, shall not cause a period 'of severance to commence if such Employee is reemployed by the Company within four (4) months after his discharge or release from such service in the armed forces or governmental agency.
- (b) In the case of a Participant who does not have any nonforfeitable right to his Employer Contribution Account, periods of service before a break-in-service will not be taken into account in computing an Eligibility Period if the break-in-service equals or exceeds the greater of five (5) consecutive (12) month periods or the Participant's period of service. Such period of service will not include any period of service disregarded under the preceding sentence by reason of prior breaks-in-service.
- 2.10 "Committee" shall mean the benefits committee referred to in Article XI appointed by the Board of Directors to manage and administer the Plan.
- 2.11 "Company" shall mean The Duriron Company, Inc., or any successor thereto, or any affiliate, subsidiary or associate corporation of the company, which shall adopt this Plan for its employees with the approval of the company. For purposes of eligibility and vesting only, company shall also include any group of business entities under common control, including but not limited to proprietorships and partnerships, or a controlled group of corporations within the meaning of Section 414 of the Internal Revenue Code and any members of an affiliated service group as defined in Section 414 (m) of the Internal Revenue Code.
- 2.12 "Compensation" of any Employee for any Plan Year shall mean all base salary, but excluding commissions, bonuses, overtime pay, reimbursements or other expense allowances, fringe benefits (cash and noncash), moving expenses, deferred compensation, and welfare benefits, paid by the Company to such Employee during such Plan Year for services rendered by the Employee to the Company as may be reflected on Form W-2 issued by the Company to the Employee, and any amounts deferred by the Employee under this qualified Plan during the Plan Year as Salary Deferral Contributions. For purposes of this Plan, compensation shall be limited to Two Hundred Thousand (\$200,000) Dollars, which amount shall be increased by the Adjustment Factor.

For purposes of applying the limitations of Section 415 of the Internal Revenue Code as set forth in Section 6.4 of the Plan or in determining if an Employee shall be deemed to be a Highly Compensated Employee as defined in Section 414(q) of the Internal Revenue Code, the term compensation shall mean a Participant's wages, salaries, fees for professional services, and other amounts for personal services actually rendered in the course of employment with the Company (including but not limited

-3-

4

CONFORMED COPY

to, commissions paid, salaries, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips and bonuses and in the case of a Participant who is an Employee within the meaning of Section 401(c) (1) of the Internal Revenue Code and the regulations thereunder, the Participant's earned income (as described in Section 401(c) (2) of the Internal Revenue Code and the regulations thereunder) paid or accrued during the limitation year. Hereunder, compensation shall exclude:

deferred compensation to the extent that, before the application of the Section 415 of the Internal Revenue Code limitations to the Plan, the contributions are not includable in the gross income of the Employee for the taxable year in which contributed;

(ii) Company contributions made on behalf of an Employee to a simplified employee pension plan described in Section 408(k) of the Internal Revenue Code to the extent such contributions are deductible by the Employee under Section 219 of the Internal Revenue Code;

(iii) any distributions from a plan of deferred compensation regardless of whether such amounts are includable in the gross income of the Employee when distributed except any amounts received by an Employee pursuant to an unfunded non-qualified plan to the extent such amounts are includable in the gross income of the Employee;

- (b) amounts realized from the exercise of a non--qualified stock option or when restricted stock (or property) held by an Employee either becomes freely transferable or is no longer subject to a substantial risk of forfeiture;
- (c) amounts realized from the sale, exchange, or other disposition of stock acquired under a qualified stock option; and
- (d) other amounts which receive special tax benefits, such as premiums for group term life insurance (but only to the extent that the premiums are not includable in the gross income of the Employee), or contributions made by the Company (whether or not under a salary reduction agreement) towards the purchase of any annuity contract described in Section 403(b) of the Internal Revenue Code (whether or not the contributions are excludable from the gross income of the Employee). For Plan Years beginning after December 31, 1988, compensation as described above shall be limited to Two Hundred Thousand (\$200,000) Dollars (unless adjusted in the same manner as permitted under Section 415(d) of the Internal Revenue Code.

In determining the compensation of a Participant for purposes of the Two Hundred Thousand (\$200,000) Dollar compensation limitation and allocating Employer discretionary contributions under Section 6.3 of the Plan, the rules of Section 414(q) (6) of the Internal Revenue Code shall apply, except in applying such rules, the term "family" shall include only the spouse of the Participant and any lineal descendants of the Participant who have not attained age nineteen (19) before the close of the Plan Year. If, as a result of the application of such rules the adjusted Two Hundred Thousand (\$200,000) Dollar limitation is exceeded, then (except for purposes of determining the portion of compensation up to the integration level if this Plan provides for permitted disparity), the limitation shall be prorated among the affected individuals in proportion to each such individual's compensation as determined under this Section prior to the application of this limitation.

- 2.13 "Determination Date" shall mean the last day of the preceding Plan Year, or, in the case of the first Plan Year, the last day of such Plan Year.
- 2.14 "Disability" shall mean the incapacity of a Participant to perform any employment which would be appropriate for a person of his prior physical status, intellectual ability and experience, due to any medically demonstrable physical or mental condition which has continued for a period of six (6) consecutive months, incapacity which qualifies the Participant to receive disability payments under the federal Social Security Act and any incapacity which the Committee finds, on the basis of qualified medical evidence, permanently shall prevent such Employee from being able to engage in any employment with the Company or in any other employment or occupation for remuneration or profit which might reasonably be considered within his capabilities, other than in such employment which is found to be for purposes of rehabilitation not incompatible with

the finding of permanent and total disability; PROVIDED, HOWEVER, that such incapacity shall not include (i) incapacity resulting exclusively from injury or disease incurred while serving in the Armed Forces of the United States and compensable as such by a federal disability allowance; (ii) incapacity caused by chronic alcoholism or addiction to narcotics; (iii) incapacity resulting from the Employee's engaging in a felonious act; or (iv) incapacity resulting from an intentionally self-inflicted injury or illness.

- 2.15 "Effective Date" shall mean January 1, 1991.
- 2.16 "Eligibility Period" shall mean a Year of Service.
- 2.17 "Eligible Employee" shall mean an Employee who is a member of the Union and who meets the participation requirements of Section 4.1 of this Plan.
- 2.18 "Employee" shall mean each individual who now or hereafter is employed by the Company, who falls under the usual common law rules applicable in determining the employer-employee relationship and is a member of the Union.
- 2.19 "Employee Salary Deferral Account" shall mean the aggregate of an Employee's Salary Deferral Contributions and the earnings thereon.
- 2.20 "Employee Matched Voluntary Contribution Account" shall mean the aggregate of an Employee's after-tax nondeductible Matched Voluntary Contributions and the earnings thereon.
- 2.21 "Employee Nonmatched Voluntary Contribution Account" shall mean the aggregate of an Employees after-tax nondeductible Nonmatched Voluntary Contributions and the earnings thereon.
- 2.22 "Employer Contribution Account" shall mean the aggregate of the Company's matching contributions which shall be made in the form of cash or Company common stock and the earnings thereon.
- 2.23 "Entry Date" shall mean the first day of January and July of each year and any other date so designated by the Plan Administrator.
- 2.24 "Fiduciary" shall mean any Person who exercises any discretionary authority or discretionary control with respect to the management or disposition of Trust assets or renders any investment advice for a fee or other compensation.
- 2.25 "Fund" or "Trust Fund" shall mean the trust fund created in accordance with the provisions herein, and all earnings, increments, and additions credited thereto. Notwithstanding anything to the contrary, the fund or trust fund shall be comprised of an Equity Fund, Fixed Income Fund, a Duriron Stock Fund, and/or any other fund selected by the Committee, to be held in a common trust maintained by the Trustee.
- 2.26 "Hardship" shall mean with respect to any Participant, an immediate and heavy financial necessity (such as the need to finance a primary residence or to provide post-secondary education for a dependent) or an emergency (such as medical expenses described in Section 213(d) of the Internal Revenue Code) which may be considered a hardship under regulations promulgated under Section 401(k) of the Internal Revenue Code.
- 2.27 "Hardship Amount" shall mean an amount determined by the Committee as being the amount required by a Participant to meet the immediate and heavy financial need created by his Hardship and which is not reasonably available to such Participant from other sources. An amount shall be considered as necessary to satisfy an immediate and heavy financial need of the Employee only if:
- (a) The Employee has obtained all distributions, other than Hardship distributions, and all nontaxable loans under all plans maintained by the Company;

- (b) All plans maintained by the Company provide that the Employee's Salary Deferral Contributions will be suspended for twelve (12) months after the receipt of the Hardship distribution;
- (c) The distribution is not in excess of the amount of an immediate and heavy financial need; and
- (d) all plans maintained by the Company provide that the Employee may not make Salary Deferral Contributions for the Employee's taxable year immediately following the taxable year of the Hardship distribution in excess of the applicable limit under Section 402(g) of the Internal Revenue Code for such taxable year less the amount of such Employee's Salary Deferral Contributions for the taxable year of the Hardship distribution.
- 2.28 "Highly Compensated Employee" shall mean any Employee or former employee as defined in Section 414(q) of the Internal Revenue Code who at anytime during the Plan Year or the twelve (12) month period immediately preceding such Plan Year was:
- (a) an Employee owning more than five (5%) percent of the outstanding stock of the Company or possessing more than five (5%) percent of the total voting power of the Company;
- (b) an Employee who had annual Compensation from the Company on a calendar year basis in excess of Seventy-Five Thousand (\$75,000) Dollars;
- (c) an Employee who had total Compensation from the Company on a calendar year basis in excess of Fifty Thousand (\$50,000) Dollars and was among the top-paid twenty (20%) percent of all Employees for that Plan Year; or
- (d) an Employee who at the time was an officer of the Company if his total annual Compensation from the Company exceeds fifty (50%) percent of the maximum dollar limitation under Section 415(b) (1) (A) of the Internal Revenue Code. For purposes of this Section 2.28(d) only, the minimum number of Employees to be treated as an officer shall be one (1) and the maximum number of Employees to be treated as officers shall be fifty (50).

Family members of Employees who own five (5%) percent or more of the Company's outstanding stock or who are one of the highest paid ten (10) Employees, shall not to be treated as separate Employees. Instead, any Compensation paid to those family members will be aggregated and treated as if it were paid to the five (5%) percent owner or top ten (10) paid Employees. In performing the Actual Deferral Percentage test and Matching Contribution Percentage test, contributions to the Plan are treated as if family members were a single individual.

As used herein, "family member" shall mean an Employee's spouse, parents and lineal ascendants or descendants and their spouses and any other individuals described in Section 414(q)(6)(B) of the Internal Revenue Code.

As used herein, "former employee" shall mean an Employee who separated from service with the Company prior to the Plan Year under consideration and was a highly compensated active Employee for either:

- (i) the year of such Employee's separation; or
- (ii) any year ending on or after the Employee'S fifty-fifth (55th) birthday.

The dollar amounts specified in subsections (b) and (c) above shall be increased by the Adjustment Factor at the same time and in the same manner as allowed under Section $415\,(\mathrm{d})$ of the Internal Revenue Code.

2.29 "Hour of Service" shall mean each hour for which an Employee is directly or indirectly paid, or entitled to payment by the Company for (a) the performance of duties during the applicable computation period; (b) each

CONFORMED COPY

7

Employee is directly or indirectly paid, or entitled to payment by the Company for reasons other than for the performance of duties, such as vacations, holidays, illness, incapacity (including Disability), jury duty, lay-off irrespective of whether the employment relationship has terminated, and (c) periods for which back pay, irrespective of mitigation of damages, is either awarded or agreed to by the Company. The same hours of service will not be credited under subparagraph (a), (b), or (c) as the case may be. These hours will be credited to the Employee for the computation period or periods to which the payment, award or agreement pertains rather than the computation period in which the award, agreement, or payment is made. Notwithstanding the foregoing the maximum number of credited hours of service for reasons other than the performance of duties shall not exceed five hundred one (501) hours during any single continuous period. The hours to be credited shall be based on the number of hours the Employee would have otherwise been credited during the period for which payment is calculated. In determining the hours of service to be credited to an Employee, the provisions of Section 2530.200b-2 of the regulations issued by the Department of Labor (which are incorporated herein by reference) shall govern. Hours of service will also be credited for any individual considered an Employee for purposes of this Plan under Section 414(n) of the Internal Revenue Code.

If a Company is a member of an affiliated service group as defined in Section 414(m) of the Internal Revenue Code, or a controlled group of corporations as defined in Section 414(b) of the Internal Revenue Code, service will be credited in accordance with the rules set forth above for any employment, for any period of time, for any member of such group. Service will also be credited for any individual required to be considered an Employee for purposes of this Plan under Section 414(n) of the Internal Revenue Code, or any Company aggregated under subsection (b) or (m) of Section 414 of the Internal Revenue Code.

Notwithstanding anything to the contrary, an hour of service shall include each hour for which an Employee is compensated by the Company, under the terms of the Bargaining Agreement or otherwise for services rendered by him to the Company. An hour of service shall also mean and include each hour for which an Employee is compensated by the Company, under the terms of the Bargaining Agreement or otherwise, on account of a period of time during which no services are rendered by him to the Company (regardless of whether the Employee shall have ceased to be an Employee) but for which compensation is required under the terms of the Bargaining Agreement or otherwise, such as, where applicable vacations, holidays, etc.

- 2.30 "Leave of Absence" shall mean any leave of absence granted by the Company without pay in accordance with reasonable standards and policies uniformly observed and consistently applied, and shall include by way of illustration and not limitation, leaves of absence granted because of illness of the Employee or his family members, vacations, and pursuit of educational or vocational study.
- 2.31 "Matched Voluntary Contributions" shall mean the first six (6%) percent of Compensation (in increments of one (1%) percent) contributed to the Plan pursuant to Section 5.1 as a Voluntary Contribution on each regular payroll date.
- 2.32 "Maximum Actual Contribution Percentage" shall mean the Actual Contribution Percentage determined by the Company with respect to such Plan pursuant to the provisions of Section 4.8 which may not be exceeded by any Participant who is deemed to be a Highly Compensated Employee with respect to such Plan Year.
- 2.33 "Maximum Tax Deferral Percentage" shall mean the Tax Deferral Percentage determined by the Company with respect to such Plan pursuant to the

provisions of Section 4.8 which may not be exceeded by any Person who is a Highly Compensated Employee with respect to such Plan Year.

- 2.34 "Nonmatched Voluntary Contributions" shall mean for any Participant already contributing six (6%) percent of his Compensation as an Employee Salary Deferral or as a Matched Voluntary Contribution pursuant to Section 5.1 of this Plan, on each regular payroll date and contributes an additional amount to the Plan up to ten (10%) percent of his compensation as a Voluntary Contribution.
- 2.35 "Normal Retirement Date" shall mean a Participant's sixty-fifth (65th) birthday. Other retirement dates shall be events of retirement as set forth in the Bargaining Agreement.

-7-

8

CONFORMED COPY

- 2.36 "Participant" shall mean any Employee or former Employee who is a member of the Union and who has qualified for participation in the Plan in accordance with the provisions hereinafter set forth.
 - 2.37 "Party In Interest" shall mean:
 - (a) Any Fiduciary.
 - (b) Any Person providing services to the Plan.
 - (c) The Company.
- $\mbox{(d)} \qquad \mbox{ Any employee organization whose members are } \\ \mbox{Participants.}$
- (e) Any direct or indirect owner of fifty (50°) percent or more of (i) the combined voting power of all classes of stock entitled to vote or of the total value of shares of all classes of stock of the Company; (ii) the capital or profits interest of any partnership; or (iii) the beneficial interest of any trust or unincorporated entity which is held directly or indirectly by Persons described in subsections (c) and (d).
- $\qquad \qquad \text{(f)} \qquad \quad \text{Any Relative of any individual described in subsections (a), (b) or (e) above.}$
- (g) Any corporation, partnership, trust or estate of which (or in which) fifty (50%) percent or more of: (i) the combined voting power of all classes of stock entitled to vote or the total value of all classes of stock of such corporation; (ii) the capital interest or profits interest of such partnership; or (iii) the beneficial interest of such trust or estate is owned directly or indirectly or held by Persons described in subsections (c), (d) or (e) above.
- (h) Any Employee, officer, director (or an individual having powers or responsibilities similar to those of officers or directors), or a ten (10%) percent or more shareholder, directly or indirectly, of a Person described in subsections (c), (d), (e) or (g) above.
- (i) Any ten (10%) percent or more partner or joint venturer (directly or indirectly in capital or profits) of a Person described in subsections (b), (c), (d), (e) or (g).
- 2.38 "Payroll Authorization" shall mean a Participant's written authorization to withhold from his wages, specified percentages which shall be as either a Salary Deferral Contribution or Matched Voluntary Contribution or Nonmatched Voluntary Contribution contributed to this Plan on his behalf.
- 2.39 "Person" shall mean any individual, partnership, joint venture, corporation, mutual company, joint stock company, trust, estate,

unincorporated organization, association or employee organization.

- 2.40 "Plan" shall mean the Savings and Thrift Plan created by this agreement and any and all amendments thereto.
 - 2.41 "Plan Administrator" shall mean THE DURIRON COMPANY, INC.
- 2.42 "Plan Year" shall mean a calendar twelve (12) month period. For purposes of the limitation year and Compensation, the computation period shall coincide with the plan year.
- 2.43 "Qualifying Employer Security" shall mean an Employer Security which is stock or a marketable obligation (as defined in Section 407(e) of ERISA.
- 2.44 "Relative" shall mean a spouse, ancestor, lineal descendant or spouse of a lineal descendant.

-8-

9

CONFORMED COPY

- 2.45 "Rollover Account" shall mean any 401(k) funds rolled over or transferred to this Plan from a prior employer's 401(k) plan pursuant to Article XV.
- 2.46 "Salary Deferral Contribution" shall mean contributions made to the Plan during the Plan Year by the Company, at the election of the Participant, in lieu of cash compensation and shall include contributions that are made pursuant to a salary reduction agreement or other deferral mechanism. Such contributions must be nonforfeitable when made and distributed only as specified in Section 8.1(f) of the Plan.

With respect to any calendar year, a Participant's salary deferral contributions shall be the sum of all Company contributions made on behalf of such Participant pursuant to any qualified plan as described in Section 401(k) of the Internal Revenue Code, any simplified employee pension cash or deferred arrangement as described in Section 402(h) (1) (B) of the Internal Revenue Code, any eligible deferred compensation plan under Section 457 of the Internal Revenue Code, and any plan as described under Section 501(a) (18) of the Internal Revenue Code, and any Company contributions made on behalf of a Participant for the purchase of an annuity contract under Section 403(b) of the Internal Revenue Code pursuant to a salary reduction agreement.

- 2.47 "Tax Deferral Percentage" shall mean the percentage derived by dividing the Participant's Salary Deferral Contribution for a given Plan Year, plus, if applicable, any additional Company contributions pursuant to Section 4.8 and any other Company contributions that would fall within the confines of Section 401(k) (3) (D) of the Internal Revenue Code and satisfy the discriminatory requirements as set forth in Regulation Section 1.401(k) -1(b) (3), the provisions of which are specifically incorporated herein by reference, by his Compensation for that Plan Year.
- 2.48 "Taxable Year" shall mean the annual accounting period upon which income is reported by the Company for Federal income tax purposes.
- 2.49 "Termination Date" shall mean either (a) the date an Employee quits, retires, is discharged, or dies, or (b) the first anniversary of the date an Employee is absent from service for any approved leave of absence or (c) the date an Employee's lay-off recall rights expire.

Notwithstanding anything to the contrary, the termination date of any Employee who is absent from work by reason of (1) the pregnancy of the Employee; (2) the birth of a child of the Employee; (3) the placement of a child in connection with the adoption of the child by the Employee; or (4) the caring for the child during the period immediately following the birth or

placement for adoption, shall not occur until the second anniversary of the date such Employee is absent for said purpose.

- 2.50 "Trust" shall mean the Savings and Thrift Trust created by this Agreement, and any and all amendments thereto or some other trust agreement entered into with Trustee. I L. the Company enters into a separate trust agreement with the Trustee, the provisions of the separate trust agreement shall take precedence over the trust provisions within this document.
- 2.51 "Trustee" and/or "Trustees" shall mean the financial institution or individual(s) appointed by the Committee.
- 2.52 "Union" shall mean THE UNITED STEELWORKERS OF AMERICA, AFL-CIO-CLC, DISTRICT 30, LOCAL NUMBER 3320.
- 2.53 "Valuation Date" shall mean the last day of each calendar quarter during the Plan Year.
- 2.54 "Voluntary Contribution" shall mean Matched Voluntary Contributions and/or Nonmatched Voluntary Contributions which are after-tax contributions made by a Participant in the Plan that are included in the Participant's gross income in the year made and subject to testing under Section 401(m) of the Internal Revenue Code.

-9-

10

CONFORMED COPY

- 2.55 "Withholding Change Date" shall mean an the first day of January, April, July, and October of each year and any other date so designated by the Committee.
- 2.56 "Year of Service" shall mean, for each Employee, a full three hundred and sixty-five (365) (or three hundred sixty-six (366), if applicable) day period measured from the date, or anniversaries thereof, he is first credited with an Hour of Service either upon initial employment or reemployment following a Break in Service.

ARTICLE III PURPOSE

The purpose of the Plan and Trust is to provide retirement savings benefits for Employees of the Company who are members of the Union and who are subject to the Bargaining Agreement. It is intended that the Plan and Trust will be for the exclusive benefit of Participants, their Beneficiaries and families, and except as otherwise provided herein, that in no event shall any part of the principal or income of the Trust Fund be paid to or revert to the Company. It is further intended that the Plan and Trust will constitute a qualified profit sharing 401(k) Plan and Trust within the meaning of Sections 401(a), 401(k), 401(m), and 501(a) of the Internal Revenue Code, as amended, and such intention shall be given great weight and consideration in the interpretation and construction of any provisions hereof.

ARTICLE IV PARTICIPATION

- 4.1 Any Employee who is a member of the Union shall be eligible for participation in the Plan on the Entry Date coincident with or next following the later of (a) the date he completes an Eligibility Period; or (b) the date he attains age eighteen (18).
- $4.2\,$ (a) Any Participant who terminates employment with the Company shall cease to participate hereunder, as of his Termination Date.
- (i) A former Participant will become a Participant immediately upon his reemployment, if such former Participant has a

nonforfeitable right to all or a portion of his Employer Contribution Account as of his Termination Date.

(ii) A former Participant who has satisfied the requirements of Section 4.1 who did not have a nonforfeitable right to any portion of his Employer Contribution Account as of his Termination Date will be considered a new Employee, for eligibility purposes, if his Break in Service equals or exceeds the greater of five (5) consecutive twelve (12) month periods or the Participant's period of service prior to such Break in Service. If such former Participant's period of service before his Termination Date may not be disregarded pursuant to the preceding sentence, such former Participant shall participate in the Plan upon the first Entry Date after his reemployment.

(iii) Any former Employee who was never a Participant and is reemployed as an Employee will be eligible to participate subject to the provisions of Section 4.1.

(b) Any Participant who transfers to a classification of Employee ineligible to participate in this Plan shall cease to participate hereunder. However, his service in such an ineligible status shall be included in determining his period of service as if such service were performed in an eligible status, and he shall not be considered to have retired or terminated employment for purposes of the Plan so long as he remains an Employee of any Company. He may resume participation in the Plan immediately upon resuming eligible employment status.

Any Employee who transfers from ineligible to a classification of Employee eligible to participate in this Plan shall commence participation in the Plan in accordance with the provisions of Section 4.1. If such Employee has already satisfied the

-10-

11

CONFORMED COPY

requirements of Section 4.1 except for his ineligible classification, he shall commence participation immediately.

- 4.3 The Company shall advise the Committee which Employees are initially eligible to participate in the Plan and which Employees are eligible on each Entry Date. In the event that any question arises as to the eligibility of any Employee, the decision of the Committee as to such Employee's eligibility shall be binding upon the Company, the Employees, the Participants, their beneficiaries, and any and all other persons having any interest hereunder.
- 4.4 Subject to such conditions as the Company shall determine, an Eligible Employee shall elect to become a Participant and shall file with the Company his authorization for the Company to withhold, and to pay to the Trustees, as the Company's Contribution to the Plan with respect to that Plan Year pursuant to the Participant's authorization to defer his salary, the amount of such Participant's Salary Deferral Contribution or Matched Voluntary Contribution and/or Nonmatched Voluntary Contribution for the period during which such Payroll Authorization shall be in effect. The permitted amount of a Participant's Salary Deferral Contribution or Matched Voluntary Contribution and/or Nonmatched Voluntary Contribution shall be based on an elected percentage of a Participant's Compensation as permitted under the provisions of Article V.
- 4.5 During each Plan Year for which a Payroll Authorization is in effect, the Company shall withhold from the Compensation of each such Participant the percentage of Compensation specified in such Participant's Payroll Authorization.
- 4.6 As of any Withholding Change Date, any Participant shall have the right to change the amount of his Salary Deferral Contribution or Matched Voluntary contribution and/or Nonmatched Voluntary Contribution and the investment of his future contributions to the Plan by notifying the Committee

or the Plan Administrator. Such notification shall be accomplished by completing a new Payroll Authorization which shall be binding upon the Company, the Employees, the Participants, their beneficiaries, and any and all other persons having any interest hereunder. As of the first day of January of each year, a Participant may authorize the change of the investment of his existing Employee Salary Deferral Account, Employee Matched Voluntary Contribution Account and Employee Nonmatched Voluntary Contribution Account in the Plan.

- 4.7 (a) Any Participant shall have the right to suspend Salary Deferral contributions or Matched Voluntary contributions and/or Nonmatched Voluntary Contributions. Such suspension must be transmitted in writing to the Committee or the Plan Administrator who shall act upon the suspension request as soon as administratively possible.
- (b) Any Participant who has suspended Salary Deferral contributions or Matched Voluntary Contributions and/or Nonmatched Voluntary Contributions to the Plan may elect to commence Salary Deferral Contributions or Matched Voluntary Contributions and/or Nonmatched Voluntary Contributions to the Plan as of any Withholding Change Date after the date he suspended Salary Deferral Contributions or Matched Voluntary Contributions and/or Nonmatched Voluntary Contributions to the Plan, by notifying the Committee or the Plan Administrator in writing at least thirty (30) days prior to the Withholding Change Date for which the election will be effective.
- 4.8 (a) The Company may amend, suspend or revoke its Payroll Authorization agreement with any Participant at any time if the Company determines that such amendment, suspension or revocation is necessary to insure that the Annual Additions to a Participant's account will not exceed the limitations of Section 6.4 or to insure that for each Plan Year the Maximum Tax Deferral Percentage and/or the Maximum Actual Contributed Percentage for the Highly Compensated Employees, as referred to in Section 5.1 shall be satisfied with respect to such Plan Year.
- (b) If, at the end of any Plan Year, after taking into account the contributions of all Participants made with respect to such Plan Year, neither of the tests referred to in Section 5.1 herein shall have been satisfied with respect to such Plan Year, then, the Company shall within thirty (30) days after the end of the Plan Year establish a Maximum Tax Deferral Percentage and a Maximum Actual Contribution Percentage for the Highly Compensated Employees. As of the Anniversary Date, the Company shall, using the leveling method described in Regulation Section 1.401(k)-1(f)(2), charge pro rata against the account Balance of each Highly Compensated Employee whose Tax Deferral Percentage or Actual Contribution Percentage shall exceed the Maximum Tax Deferral Percentage or the Maximum Actual Contribution Percentage, respectively, with respect to such Plan

-11-

12

CONFORMED COPY

Year, an amount which is necessary so that the Maximum Tax Deferral Percentage and the Maximum Actual Contribution Percentage, as established by Sections 401(k) and 401(m) of the Internal Revenue Code, respectively, shall not be exceeded by such Participant. The determination and correction of the Maximum Tax Deferral Percentage and/or the Maximum Actual Contribution Percent of a Highly Compensated Participant whose Actual Deferral Percentage and/or Actual Contribution Percentage is determined under the family aggregation rules shall be accomplished as follows:

(i) If the Actual Deferral Percentage and/or the Actual Contribution Percentage for the Highly Compensated Employee is determined in accordance with Section 5.1(c) (ii) (A) (2) and/or Section 5.1(d) (ii) (A) (2), then the Actual Deferral Percentage and/or Actual Contribution Percentage shall be reduced as required herein and the excess Salary Deferral Contributions, Matched Voluntary Contributions, Nonmatched Voluntary Contributions, or Matching Company Contributions for the family unit shall be allocated among the family members in proportion to the Salary Deferral Contributions, Matched Voluntary Contributions, Nonmatched Voluntary

Contributions, or Matching Company Contributions of each family member that were combined to determine the groups Actual Deferral Percentage and/or Actual Contribution Percentage.

If the Actual Deferral Percentage and/or the (ii) Actual Contribution Percentage for the Highly Compensated Employee is determined under Section 5.1(c) (ii) (A) (1) and/or Section 5.1(d) (ii) (A) (1), then the Actual Deferral Percentage and/or the Actual Contribution Percentage shall first be reduced as required herein, but not below the Actual Deferral Percentage and/or the Actual Contribution Percentage of the group of family members who are not Highly Compensated Employees without regard to family aggregation. The excess Salary Deferral Contributions, or excess Matched Voluntary Contributions, and/or the excess Non-matched Voluntary Contributions, and/or the excess Matching Company Contributions resulting from this initial reduction shall be allocated (in proportion to Salary Deferral Contributions, or Matched Voluntary Contributions, and/or Nonmatched Voluntary Contributions, and/or Matching Company Contributions) among the Highly Compensated Employees whose Salary Deferral Contributions, or Matched Voluntary Contributions, and/or Nonmatched Voluntary Contributions, and/or Matching Company Contributions were combined to determined the Actual Deferral Percentage and/or the Actual Contribution Percentage. If further reduction is still required, then Salary Deferral Contributions, or Matched Voluntary Contributions, and/or Nonmatched Voluntary Contributions, and/or Matching Company Contributions resulting from this further reduction shall be determined by taking into account the contributions of all family members and shall be allocated among them in proportion to their respective Salary Deferral Contributions, or Matched Voluntary Contributions, and/or Nonmatched Voluntary Contributions, and/or Matching Company Contributions. Effective retroactively to the dates as of which such contributions were made, any amounts so charged will be returned to such Participant within two and one-half (2 1/2) months of the end of the Plan Year. The amount of excess Salary Deferral Contributions, or Matched Voluntary Contributions, and/or Nonmatched Voluntary Contributions to be distributed to any Highly Compensated Employee pursuant to this Section shall be reduced by the excess Salary Deferral Contributions, or Matched Voluntary Contributions, and/or Nonmatched Voluntary Contributions previously distributed for the Taxable Year ending in the same Plan Year pursuant to Section 5.1 of the Plan, and the excess Salary Deferral Contributions, or Matched Voluntary Contributions, and/or Nonmatched Voluntary Contributions to be distributed for a Taxable Year pursuant to Section 5.1 will be reduced by the excess contributions previously distributed pursuant to this Section 4.8 for the Plan Year beginning in such Taxable Year. The income allocated to the excess Salary Deferral Contributions, or excess Matched Voluntary Contributions, and/or excess Nonmatched Voluntary Contributions, and/or excess Matching Company Contribution to be returned to the Highly Compensated Employee Participant shall be calculated in accordance with one of the methods outlined in IRS Notice 88-33, I.R.B. 1988-13, or the Proposed Regulation Sections 1.401(k) (f) (4) (ii) and/or 1.401(m)-1(e) (3) (i), respectively. The income allocated to the excess Salary Deferral Contributions, or excess Matched Voluntary Contributions, and/or excess Nonmatched Voluntary Contributions, and/or the excess Matching Company Contributions to be returned to the Highly Compensated Employee Participants, for the period between the end of the plan Year and the date of the corrective distribution, shall be calculated using the fractional method or the safe harbor method as described in Regulation Section 1.401(m)-1(e) (3) (ii) (B)&(C) which is incorporated herein by reference.

Notwithstanding anything to the contrary, forfeitures pursuant to Section 7.1 of the Plan may not be used to meet the tests prescribed in subsections (c) and (d) of Section 5.1 of the Plan.

For purposes of this Section 4.8, Matching Company Contributions shall also include Additional Company Matching

- 4.9 If any Participant shall cease to be an eligible Participant in the Plan, but continues to be employed by the Company, he shall be deemed to have suspended his Payroll Authorization as then in effect for the period during which he shall be employed as an ineligible Participant. If an ineligible Participant again becomes a Participant he shall become a Participant as of the Entry Date next following his submission of a Payroll Authorization to the Committee or the Plan Administrator.
- 4.10 The Committee shall have the right to require any Employee, at the time of his becoming a Participant, to agree in writing to be bound by the terms, covenants and conditions of the Plan and Trust.
- 4.11 The Committee, in their sole and nonreviewable discretion, may permit Employees to waive participation under this Plan. An Eligible Employee who does not wish to become a Participant must execute a written waiver stating that he has been advised by the Committee that a contribution allocation is made available to him, and so knowing, waives his right to such contribution. However, such waiver shall not preclude the Employee's participation on the Entry Date following receipt by the Committee of the Employee's written rescission of such waiver. The Committee shall maintain an executed copy of such waiver and rescission, if any.

ARTICLE V CONTRIBUTIONS

5.1 (a) Pursuant to the provisions of Article IV, Employee Salary Deferral Contributions or Employee Matched Voluntary Contributions and/or Employee Nonmatched Voluntary Contributions shall not be required of Participants as a condition of their participation in the Plan. For purposes of this Plan, all Salary Deferral Contributions shall be deemed to be derived from the Company's revenues and are therefore considered Company contributions.

[THIRD AMENDMENT REVISED SECTION 5.1(b) IN ITS ENTIRETY TO READ AS FOLLOWS, EFFECTIVE 1/1/94]

(b) (i) For each pay period, the Company shall make a Salary Deferral Contribution to the Trustee on behalf of each Participant who is employed by it during that pay period and who has elected to defer a portion of his Compensation pursuant to an effective Payroll Authorization. The Salary Deferral Contribution made on behalf of a Participant shall be in the amount (expressed as a whole percentage) by which the Participant's Compensation otherwise payable to that Participant by the Company for that pay period has been reduced pursuant to the terms of his Payroll Authorization.

(ii) In no event shall a Participant elect to defer under paragraph (i), above, a percentage of his Compensation otherwise payable to him for that Plan Year by a percentage which exceeds:

- A. 10%, reduced by
- B. the percentage of Compensation, if any, that is contributed on the Participant's behalf pursuant to paragraph (iii), below, which exceeds 6%.
- (iii) For each pay period, the Company shall make a Matched Voluntary Contribution and, if applicable, a Nonmatched Voluntary Contribution to the Trustee on behalf of each Participant who is employed by it during that pay period and who has elected to make such contribution pursuant to an effective Payroll Authorization in an amount (expressed as a whole percentage) that does not exceed:
 - A. 16%, reduced by
- B. the percentage of Compensation, if any, that is contributed on the Participant's behalf pursuant to paragraph (i), above.

- (iv) Notwithstanding anything to the contrary, a Participant's Salary Deferral Contribution for each calendar year shall not exceed \$7,000, as annually adjusted by the Secretary of Treasury.
- (c) Notwithstanding Section 6.3, with respect to the Actual Deferral Percentage for each Plan Year, either;
- (i) the Actual Deferral Percentage for the Highly Compensated Employee is not more than one and one-quarter (1 1/4) times the Actual Deferral Percentage for all other Eligible Employees; or
- (ii) the Actual Deferral percentage for the Highly Compensated Employees cannot exceed the Actual Deferral Percentage for all other Eligible Employees by more than two percentage points, and the Actual Deferral Percentage for the Highly Compensated Employees is not more than two (2) times the Actual Deferral Percentage of all other Eligible Employees. For the purpose of determining the Actual Deferral Percentage of a Highly Compensated Employee who is subject to the family member aggregation rules of Section 414 (q) (6) of the Internal Revenue Code because such Participant is either a "five percent owner" of the Company or one of the ten (10) Highly Compensated Employees paid the greatest compensation during the year, the following shall apply:
- Percentage for the family group (which shall be treated as one Highly Compensated Employee shall be the greater of: (I) the ratio determined by aggregating Salary Deferral Contributions and Compensation of all eligible family members who are Highly Compensated Employees without regard to family aggregation; and (2) the ratio determined by aggregating Salary Deferral Contributions and Compensation of eligible family members (including Highly Compensated Employees). However, in applying the Two Hundred Thousand (\$200,000) Dollar limit to Compensation, family members shall include only the affected Employee's spouse and any lineal descendants who have not attained age nineteen (19) before the close of the Plan Year.
- (B) The Salary Deferral Contributions and Compensation of all family members shall be disregarded for purposes of determining the Actual Deferral Percentage of the non-Highly Compensated Employee group except to the extent taken into account in subparagraph (A) above.
- (C) If a Participant is required to be aggregated as a member of more than one family group in a plan, all Participants who are members of those family groups that include the Participant are aggregated as one family group in accordance with subparagraphs (A) and (B) above.
- (d) Notwithstanding Section 6.3, with respect to the Actual Contribution Percentage for that Plan Year, either:
- (i) the Actual Contribution Percentage for the Highly Compensated Employees is not more than one and one-quarter (I 1/4) times the Actual Contribution Percentage for all other Eligible Employees; or
- the Actual Contribution Percentage for the (ii) Highly Compensated Employees cannot exceed the Actual contribution Percentage for all other Eligible Employees by more than two percentage points, and the Actual Contribution Percentage for the Highly compensated Employees is not more than two (2) times the Actual contribution Percentage of all other Eligible Employees. notwithstanding anything to the contrary, unless the Secretary of the Treasury prescribes an alternative, if subsection (c)(ii) is used to meet the Actual Deferral Percentage test, then the Maximum Actual Contribution Percentage must be met using subsection (d) (i). notwithstanding anything to the contrary, the multiple use limitations as set forth in Proposed Regulation Section 1.401(m)-2 shall not be violated and, if necessary, shall be corrected as set in Proposed Regulation Section 1.401(m) -2 (c) which is incorporated herein by reference. For the purpose of determining the Actual Contribution Percentage of a Highly Compensated Employee who' is subject to the family member aggregation rules of Section 414(q) (6) of the Internal Revenue Code because such Employee is either a "five percent owner" of the Company or one

of the ten (10) Highly compensated Employees paid the greatest Compensation during the year, the following shall apply:

- The combined Actual Contribution (A) Percentage for the family group (which shall be treated as one Highly Compensated Employee) shall be the greater of: (1) the ratio determined by aggregating Employee Matched Voluntary Contributions made pursuant to Section 5.1(b) (ii), Employee Nonmatched Voluntary Contributions made pursuant to Section 5.1(b) (iii), Matching Company Contributions and Additional Company Matching Contributions made pursuant to Section 5.2 and Compensation of all eligible family members who are Highly compensated Employees without regard to family aggregation; and (2) the ratio determined by aggregating Employee Matched Voluntary Contribution made pursuant to Section 5.1(b) (ii), Employee Nonmatched Voluntary Contributions made pursuant to Section 5.1(b) (iii), Matching company Contributions and Additional Company Matching Contributions made pursuant to Section 5.2 and Compensation of all eligible family members including Highly compensated Employees). However, in applying the Two Hundred Thousand (\$200,000) Dollar limit to compensation for Plan Years beginning after December 31, 1988, family members shall include only the affected Employee's spouse and any lineal descendants who have not attained age nineteen (19) before the close of the Plan Year.
- Contributions made pursuant to Section 5.1(b) (ii), Employee Nonmatched Voluntary Contributions made pursuant to Section 5.1(b) (iii), the Matching Company Contributions and Additional Company Matching Contributions made pursuant to Section 5.2, and Compensation of all family members shall be disregarded for purposes of determining the Actual Contribution Percentage of the non-Highly compensated Employee group except to the extent taken into account in subparagraph (A) above.
- (C) If a Participant is required to be aggregated as a member of more than one family group in a plan, all Participants who are members of those family groups that include the Participant are aggregated as one family group in accordance with paragraphs (A) and (B) above.
- (e) The amount of all Salary Deferral contributions and Voluntary Contributions to the Plan will be paid in cash directly to the Trustee not later than the last business day of the calendar month next succeeding the payroll date those Salary Deferral contributions and/or Voluntary Contributions were withheld. The Committee shall forward to the Trustee written instructions directing the investments of the amounts so transferred to the Trustee.
- (f) For purposes of this Section 5.1, in the event this Plan shall satisfy the requirements of Sections 401(a) (4) or 410(b) of the Internal Revenue code only if aggregated with one or more other plans, or if one or more other plans satisfy the requirements of Section 410(b) of the Internal Revenue Code only if aggregated with this Plan, then this Section shall be applied by determining the Actual Deferral Percentages and Actual contribution Percentages of Participants as if all such plans were a single plan.
- (g) For purposes of this Section 5.1 and in determining the Actual Deferral Percentage and Actual contribution Percentage of a Highly Compensated Employee Participant who is either a five (5%) percent owner of the Company or one of the ten (10) highest paid Employees of the Company, such computation shall include the Compensation and all company contributions on behalf of family members as defined in Section $414\,(q)\,(6)\,(B)$ of the Internal Revenue code.
- 5.2 Subject to the provisions of Article XIII and the Bargaining Agreement, the company shall contribute to the Trust for each Plan Year,

beginning with the first Plan Year ending on or after the Effective Date a contribution equal to twenty-five (25%) percent of the first six (6%) percent of the amount which the Participants have contributed to either their Employee Salary Deferral Accounts or to their Employee Matched Voluntary contribution Accounts (hereinafter referred to as "Matching company Contributions"). If the company and any other corporation with which it is consolidated for financial reporting purposes incurs a net loss after extraordinary items (as certified by the company's certified public accountants) for two (2) or more consecutive Plan Years, no Matching company contribution will be made for the second and immediately succeeding Plan Years in which such a loss is incurred. If the consolidated group has net income for any Plan Year for which the Company fails to make a Matching company contribution in anticipation of a net loss, the Company will make, after the Company's certified public accountants have certified the consolidated group's net income for that Plan Year, a Matching company contribution. In addition,

-15-

16

CONFORMED COPY

in any Plan Year during which the return on consolidated shareholders' equity (as reported in the company's annual shareholders report) is at least nine (9%) percent, the Company will contribute an additional matching contribution to the Plan (hereinafter referred to as "Additional company Matching Contribution"). This Additional company Matching contribution shall be based upon the following schedule:

The Company Will Make an Additional Company Matching Contribution Equal to the Following Percentage of Employee Salary Deferral Contributions or Employee Matched Voluntary Contributions Made During the Plan by Participants

If the Return on Shareholders' Equity Equals or Exceeds

9%	but	less	than	10%	5%
10%	but	less	than	12%	15%
12%	but	less	than	16%	25%
16%	or q	reate	er		35%

- 5.3 The amount of the Company's contribution to the Plan attributable to Matching Company Contributions and Additional Company Matching Contributions to the Plan for each Plan Year shall be in the form of cash or Company stock. The Matching Company Contributions shall be paid to the Trustee either in a single payment or in installments, not later than the last day of the month following the calendar quarter in which the Employee Salary Deferral Contributions or the Matched Voluntary Contributions were made. The Additional Company Matching Contributions for a Plan Year shall be paid to the Trustee as soon as possible after the Plan Year to which it reflects but in no event later than the period provided under the Internal Revenue code and other applicable laws for the payment of a deductible contribution for the Taxable Year in which such Plan Year ends. Payment by the Company of the contributions required by this Article shall completely discharge the Company's financial obligations under the terms of this Plan and the Bargaining Agreement to which it is subject.
- 5.4 The Trustee shall not be under any duty to inquire into the correctness of the amounts contributed and paid over to them pursuant to this Article, nor shall the Trustees be under any duty to enforce payment of any contribution to be made hereunder by the Company.
- 5.5 Notwithstanding anything to the contrary, the Committee shall be charged with the duty of ascertaining the correctness of amounts contributed and paid over to the Trustee pursuant to this Article. should the Committee ascertain that amounts so contributed are incorrect and if upon notification of

such error, the Company does not agree or fails to rectify the error the issue shall be submitted for arbitration.

ARTICLE VI ACCOUNTS AND ALLOCATIONS

- 6.1 The Trustee shall maintain, or cause to be maintained, such accounts as may be necessary or desirable to achieve the purposes of the Plan.
- 6.2 The Trustees shall maintain, or cause to be maintained, a separate Employee Salary Deferral Account, Employee Matched Voluntary Contribution Account, Employee Nonmatched Voluntary Contribution Account, an Employer contribution Account and Rollover Account for each Participant.
- 6.3 (a) The Matching company contributions to the Plan for any Plan Year shall be deemed to have been made as of the last day of each quarter of the Plan Year, and shall be allocated to the Trust Fund and apportioned among and credited to the accounts of those Participants who have made contributions to their Employee Salary Deferral Account or Employee Matched

-16-

17

CONFORMED COPY

Voluntary contribution Account during such quarter during the Plan Year. The apportionment shall be in the ratio that the first six (6%) percent Salary Deferral Contributions or Matched Voluntary contributions of each such Participant for the quarter bears to the first six (6%) percent of the Salary Deferral Contributions or Matched Voluntary Contributions for all such Participants for such quarter. Notwithstanding anything to the contrary, in allocating the Matching company Contributions, only the first six (6%) percent of a Participant's Compensation contributed to the Plan as either Salary Deferral Contributions or Matched Voluntary contribution shall be taken into account.

- The apportionment of the Additional Company Matching (b) Contributions shall be deemed to have been made as of the last day of the Plan Year, and shall be allocated to the Trust Fund and apportioned among and credited to the accounts of those Participants who are eligible for Matching Company contributions and who are employed with the Company on said date; PROVIDED, HOWEVER, if a Participant terminated employment during the Plan Year after having attained his retirement date as set forth in the Bargaining Agreement, or due to his Disability or death, he shall, if applicable, share in the Additional Company Matching contribution for such Plan Year. Such apportionment shall be in the ratio that the first six (6%) percent Salary Deferral contributions or Matched Voluntary Contributions of each such Participant for the Plan Year bears to the first six (6%) percent of the Salary Deferral Contributions or Matched Voluntary Contributions for all such Participants for such Plan Year. Notwithstanding anything to the contrary, in allocating the Additional company Matching Contributions, only the first six (6%) percent of a Participant's compensation contributed to the Plan as either Salary Deferral Contributions or Matched Voluntary Contribution shall be taken into account.
- 6.4 (a) With respect to any Participant for any Plan Year, the sum of:
- (i) The contribution made by the Company on behalf of such Participant (which amounts shall include any Salary Deferral contributions and/or Voluntary Contributions) for such Plan Year; plus
- (ii) If applicable, the Participant's pro rata share of forfeitures apportioned to the Participant's account for such Plan Year, allocated to the Participant under this or any other defined contribution plan of the Company which is qualified under Section 401(a) of the Internal Revenue Code, shall not exceed the lesser of: (A) Thirty Thousand (\$30,000) Dollars, or if greater, one-quarter (1/4) of the defined benefit dollar limitation as set forth in Section 415(b) (I) of the Internal Revenue code as

in effect for the limitation year; or (B) twenty-five (25%) percent of the Participant's Compensation for such Plan Year (such limitation being hereinafter referred to as the "Maximum contribution Limit").

For purposes of this Section, compensation shall exclude any Salary Deferral Contributions.

(b) If the Maximum Contribution Limit is exceeded, then a portion of the Participant's share of forfeitures for such Plan Year shall be reduced by the amount of such excess. If the Maximum Contribution Limit is still exceeded after a reduction of the Participant's share of forfeitures, then the Company's contributions which would have been allocated to the Participant under Section 6.3 shall be reduced. The reduction shall be equal to the pro rata portion of the contribution allocable to the Participant under this Plan divided by the total contribution allocable to the Participant under all defined contribution plans of the Company. The excess amounts shall be allocated and reallocated to other Participants in the Plan pursuant to Regulation Section 1.415-6(b)(6)(i).

If an excess still exists, the excess will be held unallocated in a suspense account. This suspense account will be applied to reduce future Company contributions for all remaining Participants in the next Plan Year, and each succeeding year if necessary. The suspense account shall not be credited with any investment gains and losses of the Trust Fund. Upon termination of the Plan, unallocated amounts in the suspense account shall, notwithstanding any other provisions of the Plan, revert to the Company.

(c) Any Matching company Contribution or Additional Company Matching Contribution which cannot be fully allocated to a Participant's Employer contribution Account without violating the limitations on Annual Additions as set forth in Section 6.4(a), shall be paid to the Participant on whose behalf the contribution was made under Article V, as part of his regular

-17-

18

CONFORMED COPY

Compensation.

(d) For any Participant who is or has been a Participant in a defined benefit plan maintained by the company whether or not terminated, or a Participant in a welfare benefit fund (as defined in Section 419(e) of the Internal Revenue Code) or an individual medical account (as defined in Section 415(l) (2) of the Internal Revenue Code) maintained by the Company, the rate of accruals under the defined benefit plan shall first be reduced, for any Plan Year, to the extent necessary to prevent the sum of the following fractions, computed as of the close of the Plan Year, from exceeding one (1.0):

Defined Benefit Plan Fraction

Numerator: The sum of the Participant's projected annual benefits under all of the qualified defined benefit plans maintained by the Company, whether or not terminated.

Denominator: The lesser of (a) one hundred and forty (140%) percent of the average of the highest three (3) Plan Years of annual compensation (as defined in Regulation Section 1. 415-2 (d) (8)), or (b) one hundred twenty-five (125%) percent of the dollar limitation in effect for the Plan Year under Section 415(b) (1) (A) of the Internal Revenue Code.

Defined Contribution Plan Fraction

Numerator: The sum of the aggregate amount of Annual Additions to the Participant's account for all Plan Years under all of the qualified defined contribution plans maintained by the company, whether or not terminated.

Denominator: The lesser of (a) thirty-five (35%) percent of the Participant's annual compensation (as defined in Regulation Section 1.415-2 (d) (8)) limitation for the Plan Year multiplied by all years of employment with the Company; or (b) one hundred and twenty-five (125%) percent of the dollar limitation for the Plan Year under Section 415(c) (1) (a) of the Internal Revenue Code multiplied by all years of employment with the Company. Notwithstanding anything to the contrary, the company may, in calculating the defined contribution and defined benefit plan fractions, elect to apply the transitional rules.

- (e) The amount of any such reduction shall be placed in a suspense account as set forth in subsection (b).
- 6.5 The assets of the Trust Fund shall be valued by the Trustees at their fair market value as of the close of business on each Valuation Date. The Trustees' determination of the value of the assets shall be final and conclusive for all purposes of the Plan.
- 6.6 The amount which shall be credited to the account of each Participant on each Valuation Date shall be:
- $\hbox{ (a)} \qquad \quad \hbox{The Account Balance on the preceding Valuation Date;} \\$
- (b) Such Participant's Salary Deferral Contributions or Matched Voluntary Contributions and/or Nonmatched Voluntary Contributions and allocated portion of any Matching Company Contributions and Additional Company Matching Contributions made with respect to the current Valuation Date; minus
- (c) The amount of any distributions made to the Participant, or his Beneficiary, since the preceding Valuation Date; plus
- (d) (i) With respect to a Participant's Employer Contribution Account, such Participant's pro rata share of the increase or decrease since the last Valuation Date in the fair market value of the portion of THE DURIRON STOCK FUND. Such increase or decrease in value shall be allocated to each such Participant on the current Valuation Date in the ratio of such Participant's Employer Contribution Account on the last Valuation Date to the total of the pertinent Employer contribution

-18-

19

CONFORMED COPY

Account's on the last Valuation Date of all Participants who are such on the current Valuation Date; plus

- (ii) With respect to a Participant's Employee Salary Deferral Account, Employee Matched Voluntary Contribution Account, Employee Nonmatched Voluntary Contribution Account and, if applicable, Rollover Account, such accounts shall be valued separately based on its investment in the Fixed Income Fund, the Equity Fund or any other fund selected by the Committee as provided for in Article -I and the increase or decrease therein since the last Valuation Date shall be added or subtracted, as the case may be, from his Employee Salary Deferral Account, Employee Matched Voluntary Contribution Account, Employee Nonmatched Voluntary Contribution Account and, if applicable, Rollover Account, on the preceding date as adjusted pursuant to the terms of subparagraphs (a), (b), and (c) above.
- 6.7 The Account Balance of any Participant who has terminated his employment (hereinafter referred to as the "terminal balance" of each such account) shall be the sum of (a) his Account Balance as of the Valuation Date coincident with or preceding the date of distribution of such terminal balance and (b) his Salary Deferral contributions and/or Voluntary Contributions and Matching Company Contributions and Additional Company Matching contributions to the Plan since the last Valuation Date.

VESTING OF INTERESTS OF PARTICIPANTS

7.1 The interest of a Participant in his Employee Salary Deferral Account, Employee Matched Voluntary contribution Account, Employee Nonmatched Voluntary contribution Account and Rollover Account shall at all times be one hundred (100%) percent vested and nonforfeitable. The interest of a Participant in his Employer Contribution Account shall be vested on his termination of participation in accordance with the following table. Such vesting shall be based upon the number of Years of Service of the Participant on his Termination Date, or, if his employment has not terminated, on the date his Break in Service occurs. A Participant shall be credited with Years of Service beginning with his date of hire.

PARTICIPANT'S
NONFORFEITABLE
YEARS OF SERVICE PERCENTAGE

less than 5 0
5 or more 100%

- 7.2 For purposes of this Article VII, Years of Service prior to a Break in Service and Years of Service after a Break in Service shall be aggregated in determining a Participant's vested interest in accordance with the following rules:
- (a) If the Participant had any vested interest in his Employer contribution Account as of the Termination Date, then his period of service prior to his Break in Service and his Years of Service after such Break in Service shall be aggregated if the Participant has at least one Year of Service after such Break in Service.
- (b) If the Participant had no vested interest in his Employer contribution Account as of the Termination Date, then his period of service prior to such Break in Service and his period of service after such Break in Service shall be aggregated only if the period of such Break in Service is less than the greater of (i) the period of service prior to such Break in Service or (ii) five (5) consecutive full years measured from a Participant's Termination Date.
- (c) A Participant's vested interest in his Account Balance as of the commencement of said Break in Service shall be nonforfeitable. If a Participant returns to the employ of the company after a Break in Service, his previous nonforfeitable account, if any, shall not be reinstated or increased as a result of service after said Break in Service, except as otherwise provided for in Section 8.1.
- 7.3 Any non-vested portion shall not be treated as a forfeiture until the Anniversary Date of the Plan Year in

-19-

20

CONFORMED COPY

which the Participant has incurred a Break in Service and received a distribution from the Plan, or the occurrence of five (5) consecutive one (I) Year Breaks in Service. Any and all forfeitures occurring pursuant to this Article shall first be used to reinstate prior forfeitures of Participants who are reemployed by the Company and then to reduce the Company's contributions to the Plan for the Plan Year.

- 7.4 Years of Service notwithstanding, one hundred (100%) percent of the amount credited to a Participant's Employer contribution Account shall become vested and nonforfeitable upon an Employee's termination of employment:
- (a) after reaching his retirement date as set forth in the Bargaining Agreement;

- (b) upon his death; or
- (c) by reason of his Disability.
- 7.5 Whenever the Plan's schedule of nonforfeitable percentages is amended, a Participant's vested portion of his Account Balance under the amended schedule shall not be less than the vested portion of his Account Balance computed under the prior schedule. Upon termination of employment, a Participant shall be entitled to the greater of the two (2) vesting schedules.

ARTICLE VIII DISTRIBUTIONS

- 8.1 (a) If a Participant shall terminate his employment other than by reason of attainment of his retirement date as set forth in the Bargaining Agreement, death or Disability, the Trustees at the direction of the committee shall distribute out of the Fund to such Participant, subject to subsections (b) and (c) of Section 8.1, cash and/or company stock representing the portion of the terminal balance of his account which shall be vested in him on the Valuation Date immediately prior to the date the distribution occurs to such Participant.
- (b) (i) Upon the written request of any Participant who attains his retirement date as set forth in the Bargaining Agreement, becomes disabled, or terminates employment with the Company, the Trustees shall cause to be paid to such Participant an amount in cash and/or Company stock equal to such Participant's Account Balance as of the Valuation Date immediately preceding the distribution to the Participant. Such payment shall be made as soon as administratively possible; PROVIDED, HOWEVER, distribution shall be made no later than sixty (60) days after the end of the later of: (a) the Plan Year in which he terminates his employment with the company; or (b) the Plan Year in which the Participant attains his retirement date as set forth in the Bargaining Agreement. Notwithstanding anything to the contrary, a Participant may request that the Committee leave his Account Balance in the Plan until he attains the age of seventy and one-half (70-). Any discretion exercisable under this Article of the Plan by the Committee shall be exercised in a nondiscriminatory manner, treating like Participants in like circumstances in a like manner.
- (ii) Notwithstanding the above, unless the Employee is not a five (5%) percent owner of the company, at any time during or after the calendar year in which he attained the age of sixty-six and one-half (66 1/2), and attained age seventy and one-half (70 1/2) prior to January 1, 1988, an Employee must commence distribution of his terminal balance by April 1st of the calendar year following the end of the calendar year in which he attains the age of seventy and one-half (70 1/2) whether or not he has ceased active employment.
- (c) Upon distribution, payment shall be made in one or more of the methods provided in Section 8.3 as the Participant or his beneficiary shall determine; PROVIDED, HOWEVER, no distribution in excess of Three Thousand Five Hundred (\$3,500) Dollars (including both Company and Employee contributions) or, if the Participant's Account Balance in the Plan ever exceeded Three Thousand Five Hundred (\$3,500.00) Dollars, may be made without the consent of the Participant.

-20-

21

CONFORMED COPY

(d) A Participant who terminates his employment and receives a distribution of his vested interest which represents less than one hundred (100%) percent of his Employer contribution Account Balance and is subsequently rehired prior to five (5) consecutive one (1) year Break in Service, shall have his Employer contribution Account balance restored to the amount standing to his credit as of the date of his termination of employment,

unadjusted by any subsequent gains or losses without having to repay the amounts distributed to him from the Plan.

- (e) In the case of a Participant who has a Break in Service of more than five (5) years, all service after such Break in Service will be disregarded for purposes of vesting in the Participant's Employer Contribution Account prior to such Break in Service. Such Participant's service prior to his Break in Service will count in vesting of his Employer Contribution Account only if:
- (i) such Participant has any nonforfeitable interest in his Employer Contribution Account attributable to contributions at the time of separation from service; or
- (ii) upon returning to service, his Break in Service is less than the period of service prior to the Break in Service.
- (f) (i) Salary Deferral Contributions, qualified non-elective contributions, and qualified matching contributions, and income allocable to each are not distributable to a Participant or his beneficiary or beneficiaries, in accordance with such Participant's or beneficiary or beneficiaries election, earlier than upon separation from service, death, or Disability. Such amounts may be distributed upon:
- $\hbox{(A)} \qquad \hbox{Termination of the Plan without the establishment of another defined contribution plan;}$
- (B) The disposition by a corporation to an unrelated corporation of substantially all of the assets (within the meaning of Section 409 (d) (2) of the Internal Revenue Code) used in a trade or business of such corporation if such corporation continues to maintain this Plan after the disposition, but only with respect to Employees who continue employment with the corporation acquiring such assets;
- (C) The disposition by a corporation to an unrelated entity of such corporation's interest in a subsidiary (within the meaning of Section 409(d) (3) of the Internal Revenue code) if such corporation continues to maintain this Plan, but only with respect to Employees who continue employment with such subsidiary; or
- (D) The attainment of age fifty-nine and one-half (59 1/2).
- (ii) (A) If the Plan Committee approves, Employee Nonmatched Voluntary Contributions and earnings (to the extent not subject to a qualified domestic relations order) may be withdrawn without penalty effective as of the end of any calendar quarter, if notice of that withdrawal has been given at least thirty (30) days before the effective date of that withdrawal.
- (B) If the Plan Committee approves, a Participant may withdraw his Matched Voluntary Contributions and earnings (to the extent not subject to a qualified domestic relations order without penalty) once during each Plan Year but only if he also has withdrawn the balance in his Nonmatched Voluntary contributions. These withdrawals are effective as of the end of any calendar quarter, if notice of that withdrawal has been given at least thirty (30) days before the effective date of that withdrawal.
- (C) If the Committee approves, a Participant who is fully vested in the Plan (has completed at least five (5) Years of Service) and who already has withdrawn the balance of his Employee Matched and Nonmatched Voluntary contribution Accounts, may withdraw funds from his Employer contribution Account that have been held

subject to a qualified domestic relations order). The two (2) year restriction is waived if the Participant has made Matched Voluntary contributions for at least five (5) years ending on the date of withdrawal. These withdrawals are effective as of the end of any calendar quarter if notice of that withdrawal has been given at least thirty (30) days before the effective date of that withdrawal. If a Participant makes a withdrawal under this subsection, he may not make any Employee Salary Deferral contributions and Voluntary Contributions until the calendar quarter six (6) months after that withdrawal.

(D) If the Plan Committee approves, a Participant who already has withdrawn the balance of his Employee Matched Voluntary contribution Account, Employee Nonmatched Voluntary contribution Account and Employer contribution Account may withdraw his Salary Deferral Contributions, prior 401(k) rollovers in his Rollover Account, plus earnings (to the extent not subject to a qualified domestic relations order) but not more frequently than once each year if the Participant has attained the age fifty-nine and one-half (59 $^{\sim}$) or older. If a Participant makes a withdrawal under this subsection, he may not make Voluntary Contributions or Employee Salary Deferral contributions until the calendar quarter six (6) months after that withdrawal.

(iii) Notwithstanding anything to the contrary, if the Plan Committee approves, a Participant who has already withdrawn the balance of his funds under subsection (ii) (A), (ii) (B), and (ii) (C) above, upon meeting the requirements of Hardship as defined in Section 2.26, the Trustees at the direction of the Committee shall cause to be paid to such Participant an amount in cash equal to the lesser of:

(A) the amount specified in such

request;

(B) such Participant's cumulative Employee Salary Deferral contributions to the Plan as of the Valuation Date preceding the Participant's request; or

- (C) such Participant's Hardship Amount. Such payment shall be made as promptly as practicable after such request is approved by the Committee, and the Trustees shall charge such payment against the Participant's Employee Salary Deferral Account as of the Valuation Date preceding the Participant's request. Notwithstanding anything to the contrary, any Participant receiving a Hardship distribution pursuant to this subsection shall automatically be suspended from making Salary Deferral contributions and Voluntary Contributions to the Plan for a period of twelve (12) months from the date of the Hardship distribution.
- (g) All withdrawals pursuant to subsection (f) must be in cash and may be requested; (i) as one hundred (100%) percent of the available amount; or (ii) in specified whole dollar increments.
- (h) Upon the Committee's receipt of a distribution request as the result of a qualified domestic relations order within the meaning of Section 414(p) of the Internal Revenue code, the amount paid to the alternate payee shall be the amount specified in the order.
- (i) For purposes of this Section 8.1, if the value of an Employee's vested Employer Contribution Account is zero (0), the Employee shall be deemed to have received a distribution of such vested Employer contribution Account.
- 8.2 If a Participant shall terminate his employment after having attained his retirement date as set forth in the Bargaining Agreement, or due to his death or Disability, at the direction of the Committee, the Trustees shall cause to be paid or distributed to such Participant, or his beneficiary, as the case may be, cash and/or Company stock, representing the terminal balance of his account. If distribution shall be made by reason of the retirement, death, or Disability of a Participant, such distribution shall be made in one or more of the methods provided in Section 8.3, as the Participant, in his sole discretion, shall determine.
- 8.3 Distributions to a Participant under Section 8.2 can only be made:

- $\hbox{(a)} \qquad \qquad \hbox{In an immediate lump sum payment in cash and/or company stock;}$
- (b) In a deferred lump sum payment in cash and/or company stock; or
- (c) In the form of fixed annual installments, for a period not to exceed the life expectancy of the Participant and his designated beneficiary. Each periodic payment to the Participant's beneficiary will be no greater than each periodic payment made during the lifetime of the Participant. The present value of payments to be made to the Participant shall be more than fifty (50%) percent of the present value of the total payments to be made to the Participant and his beneficiary determined as of the date benefits commence. If this method of distribution is selected, then the Trustees shall segregate the Participant's terminal balance from the other assets of the Trust Fund and shall deposit such terminal balance in an investment vehicle selected by the committee, and any interest earned thereon shall be distributed to the Participant, or his beneficiary, at least annually.
- 8.4 Notwithstanding anything to the contrary, if a Participant shall terminate his employment prior to completion of five (5) years of service with the Company, prior to his retirement date as set forth in the Bargaining Agreement and not as a result of death or Disability, distribution to the Participant shall only be made as soon as administratively feasible and only in the form of an immediate lump sum payment. If a Participant shall terminate his employment after completion of five (5) Years of Service with the Company but not as a result of his attainment of his retirement date as set forth in the Bargaining Agreement, death, or Disability, distribution to the Participant can only be made in the form of an immediate lump sum distribution or at the end of the Plan Year of termination.
- 8.5 Notwithstanding any provision to the contrary, this Article shall comply with Section 401(a) (9) of the Internal Revenue code and the Regulations issued thereunder. For purposes of this Article, the life expectancy of a Participant and his spouse shall not be recalculated unless the Trustees specifically consent to such redetermination in writing.

[SECOND AMD ADDED SECTION 8.6, EFFECTIVE 1/1/93]

Notwithstanding any provision of the Plan otherwise limiting the following provisions of this subsection, on and after January 1, 1993, any distributee (as defined below) under the Plan may elect, at such time and in such manner as the Committee shall provide, to have any amount that is otherwise payable to him under the Plan and that constitutes an "Eligible Rollover Distribution" paid in a "Direct Rollover" to an "Eligible Retirement Plan" (each as defined below) specified by the Distributee. The term "Eligible Rollover Distribution" means any payment of all or a portion of a Participant's Accounts except a payment that is not includible in the payee's gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities) or is required to be made under the provisions of section 401(a)(9) of the Code. For purposes of this subsection, the term "Distributee" means a Participant, his surviving spouse and his spouse or former spouse if an alternate payee. The term "Direct Rollover" means, with respect to any Distributee, a payment by the Trustee to the Eligible Retirement Plan specified by the payee. The term "Eligible Retirement Plan" means an individual retirement account, an individual retirement annuity, an annuity plan or a qualified trust (as described in sections 408(a), 408(b), 403(a) and 401(a) of the Code, respectively), that accepts a Distributee's Eligible Rollover Distribution; provided, however, that, in the case of a surviving spouse, the term Eligible Retirement Plan includes only an individual retirement account or an individual retirement annuity.

Each Participant must designate, upon such form filed with the Committee as they shall prescribe, a Person to whom, in the event of his death, his interest or any undistributed balance thereof shall be paid. If a Participant is married at the time of his death, then the Participant shall be deemed to have designated his spouse as beneficiary.

Each Participant shall have the right with the specific written consent of his spouse if he is married, to designate a

-23-

24

CONFORMED COPY

beneficiary other than his surviving spouse, to receive the benefits provided by the Plan in the event of his death. Such election must be witnessed by an authorized representative of the Plan or a notary public and must acknowledge the spouse's understanding of the effect of such election. The beneficiary so designated may be changed at any time and from time to time prior to completion of distribution of the Participant's interest by his filing with the committee, with the consent of his spouse if he is married, upon such form as they shall prescribe, a notification of change of beneficiary.

If no person shall be designated as such beneficiary, or if the designated beneficiary shall not survive the Participant, payment of his interest or of any undistributed balance thereof shall be made to his spouse, if living, and if not, then to his living issue, per stirpes, and if there be none, then to his living parents, and if there be none, then to his estate.

ARTICLE X

- 10.1 There shall be a Trustee selected by the Committee. Vacancies arising by reason of resignation, removal, or otherwise, shall be filled by the Committee. Any Trustee may resign of his own accord by delivering his written resignation to the Committee. Additional Trustees may be designated by the Committee from time to time.
- 10.2 The Trustee shall discharge its duties with respect to the Plan solely in the interest of the Participants and their beneficiaries and for the exclusive purpose of providing benefits to the Participants and their beneficiaries and defraying reasonable expenses of administering the Plan. The Trustees shall act with the care, skill, prudence and diligence under the circumstances then prevailing that a prudent man acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character and with like aims by diversifying the investments of the Trust so as to minimize the risk of large losses, unless under the circumstances it is clearly prudent not to do so.

ARTICLE XI ADMINISTRATION OF PLAN

- 11.1 The Committee shall be charged with the administration of the Plan, and shall possess all of the powers necessary to enable it to carry out its duties in that regard. The Committee and the Plan Administrator shall have the power to interpret or construe the Plan; to determine all questions that may arise as to the eligibility, status, rights and obligations of the Participants and others; and to decide any dispute arising hereunder. The committee shall also have such powers with respect to the Fund as are hereinafter set forth in Article XI.
- 11.2 The committee is authorized to make such rules and regulations as they may deem necessary to carry out the provisions of the Plan; to employ attorneys, accountants, investment advisors, specialists and other such agents as they shall deem necessary or desirable; and to direct the committee regarding such rules and regulations. The Committee shall have the authority to appoint an investment manager or managers to manage the assets of the Fund and such appointment may include the power to acquire and dispose of such assets. The Committee may charge the compensation of such attorneys,

accountants, investment advisors, investment managers, specialists and other agents and any other expenses against the Fund.

- 11.3 The Committee may adopt such by-laws and regulations as they deem desirable for the conduct of their affairs. They may appoint a secretary and such other officers as they shall deem advisable. The Committee shall act by majority and such action may be taken either by a vote at a meeting or in writing without a meeting. The committee may by such majority action appoint subcommittees and may appoint any one or more of themselves or an agent to execute any document or documents, or to take any other action, including the exercise of discretion, on behalf of the committee.
- 11.4 The committee, shall serve without compensation for their services as such. If the committee members are full-time Employees of the Company, they shall serve without compensation for their services.
- 11.5 The Committee shall keep, or cause to be kept, all such books of account, records and other data as may be

-24-

25

CONFORMED COPY

necessary or advisable in their judgment for the administration of the Plan, the proper reflection of the affairs thereof, and the determination of the amount of the vested and/or nonvested interests of the respective Participants thereof. The Committee shall keep on file, in such form as they shall deem convenient and proper, all annual reports of the Fund. Any Participant shall have the right to inspect the accounts showing his interest in the Fund at a time and place designated by the Committee and the Trustee.

- 11.6 The Committee may consult with legal counsel (who may be of counsel to the Company), concerning any question which may arise with reference to their duties under this Agreement, and if counsel has been approved by the Company, the opinion of such counsel shall be full and complete protection in respect of any act ion taken or suffered by the Committee hereunder in accordance with the opinion of such counsel.
- 11.7 Any action by the Board of Directors pursuant to any of the provisions of this Agreement shall be evidenced by a resolution of the Board of Directors certified to the Committee over the signature of the Secretary of the Company, and the committee shall be fully protected in acting in accordance with any and all such resolutions so certified to them.
- 11.8 The Committee, in making any payments, taking any action, or making any decisions with respect to any matter may rely conclusively upon the authenticity and contents of certificates, directions, statements, resolutions, regulations, requisitions and communications from the Company or the Board of Directors, furnished or delivered to them as herein provided.
- 11.9 Any Participant or beneficiary ("claimant") may make a request for Plan benefits by delivering a written notice to a Committee on which a claim for benefits is made. The notice shall set forth the name of the claimant, and the type of benefit for which the claim is made. The Participant shall be responsible for all fees incurred with respect to this request.

Within ninety (90) days of receiving the written notice, the Committee must inform the claimant of whether the claim will be satisfied or wholly or partially denied. If there is a denial, the notice shall set forth the following:

- (a) The specific reason or reasons for the denial.
- (b) Specific reference to the pertinent Plan provisions on which the denial is based.
- (c) A description of any material and information necessary to perfect the claim and an explanation of why such material and

information is necessary.

 $\mbox{(d)} \qquad \mbox{ An explanation of the Plan's claim review} \\ \mbox{procedure.}$

Within sixty (60) days of receiving a notice denying a claim, wholly or partially, the claimant may appeal such denial to the Committee for review. The request for a review must be in writing and contain the reasons for which such claim is made. The claimant may review all documents regarding the Plan for purposes of preparing his request for a review. The Committee shall review the appeal of the claimant and, if necessary, hold a hearing at which the claimant may present his reasons for objecting to the denial of the claim.

The Committee must render a decision regarding the appeal of the claimant not later than sixty (60) days after a claimant's request for review was received. However, if there is a need to hold a hearing, a decision is to be rendered not later than one hundred twenty (120) days after the receipt of the request for review.

The decision on review of the Committee must be in writing and must include specific reasons for the decision and is to make reference to the pertinent Plan provisions on which the decision is based.

11.10 The company may, if it so desires, purchase insurance to cover potential liability of the Committee and any other Plan Fiduciary employed by the Company by reason of the act or omission of any or all of said Committee and any other Plan Fiduciary employed by the Company. Should the Company fail or refuse to purchase such insurance, the committee and any

-25-

26

CONFORMED COPY

other Plan Fiduciaries employed by the Company, or any of them, may do so, but no assets of the Fund may be used to purchase such insurance.

11.11 The Company shall indemnify and hold harmless the committee and any other Plan Fiduciary employed by the company from and against any and all claims, losses, damages, expenses (including reasonable attorneys' fees approved by the company), and liability (including any reasonable amounts paid in settlement with the Company's) arising from any act or omission of the Committee and any other Plan Fiduciary employed by the Company, except when the same is judicially determined to be due to the willful misconduct of the Committee and any other Plan Fiduciary employed by the company.

ARTICLE XII TRUST FUND

- 12.1 The Fund shall consist of all contributions made by the company, and the earnings, increments and additions thereto.
- 12.2 (a) Except as otherwise provided in Article XVI, the Trustees are expressly authorized and empowered to invest and reinvest the Fund in such stocks (of any classification), bonds, options and other securities or other property, whether real or personal, subject to Section 10.2.
- (b) Subject to Section 10.2, the Trustees, in their discretion, may retain such amount of the Fund in the form of cash and unproductive of income as they may deem advisable. The Trustees shall not be required to pay interest on such cash balance or on cash in their hands pending investment.
- 12.3 Subject to the provisions of Section 12.4 and Article XVI, the Trustees are authorized and empowered in their discretion, but not by way of limitation:
- (a) To sell, exchange, convey, transfer or dispose of and also to grant options with respect to any property, whether real or personal,

at any time held by them, and any sale may be made by private contract or by public auction, and for cash or upon credit, or partly for cash and partly upon credit, as the Trustees may deem best, and no person dealing with the Trustees shall be bound to see to the application of the purchase money or to inquire into the validity, expediency or propriety of any such sale or other disposition;

- (b) To retain, manage, operate, repair and improve and to mortgage or lease for any period any real estate held by them;
- (c) To vote in person or by proxy on any stocks, bonds or other securities held by them; to exercise any options appurtenant to any stocks, bonds or other securities for the con-version thereof into other stocks, bonds or securities, or to exercise any rights to subscribe for additional stocks, bonds or other securities and to make any and all necessary payments therefore; to join in, or to dissent from, and to oppose, the reorganization, recapitalization, consolidation, liquidation, sale or merger of corporations or properties in which they may be interested as Trustees, upon such terms and conditions as they may deem wise;
- (d) To make, execute, acknowledge and deliver any and all deeds, leases, assignments and instruments;
- (e) With the approval of the Board of Directors to borrow or raise moneys for the purposes of the Trust to the extent that the Trustees shall deem desirable and upon such terms and conditions as the Trustees in their absolute discretion shall deem desirable or proper, and for any sum so borrowed to issue their promissory note as Trustees and to secure the repayment thereof by pledging all or any part of the Fund; and no person lending money to the Trustees shall be bound to see the application of the money loaned or to inquire into the validity, expediency or propriety of any such borrowing;

-26-

27

CONFORMED COPY

- (f) To cause any investments from time to time held by them to be registered in, or transferred into, their names as Trustees or the name of their nominee or nominees, or to retain them unregistered or in form permitting transferability by delivery, but the books and records of the Trustees shall at all times show that all such investments are part of the Fund;
- (g) To organize and incorporate (or to participate in the organization or incorporation of), under the laws of any state they may deem advisable, a holding corporation for the purposes of acquiring and holding title to any property which the Trustees are authorized to acquire under this agreement and with the power to exercise with respect thereto any or all of the functions and duties set forth in this Agreement;
- (h) To acquire and hold Qualifying Employer Securities subject to the limitations of Section $12.4\,(a)$ (v);
- $\,$ (i) $\,$ To do all acts whether or not expressly authorized which they may deem necessary or proper for the protection of the property held hereunder.
 - 12.4 A Fiduciary shall not:
- (a) Cause the Trust to engage in a transaction if they know or should know that such transaction constitutes a direct or indirect:
- $\hbox{(i)} \qquad \qquad \text{Sale or exchange, or leasing, of any property} \\ \text{between the Trust and a Party In Interest;}$
 - (ii) Lending of money or other extension of credit

between the Trust and a Party In Interest;

(iii) Furnishing of goods, services or facilities between the Trust and a Party In Interest;

 $\,$ (iv) Transfer to, or use by or for the benefit of, a Party in Interest, of any assets of the Trust; or

- (v) Acquire on behalf of the Trust, any Employer Security unless (A) the said securities are deemed to be Qualifying Employer Securities and (B) the aggregate fair market value of all Qualifying Employer Securities held by the Trust does not exceed ten (10%) percent of the fair market value of the assets of the Trust immediately after the acquisition, except as otherwise provided for in Section 407(b) of the Employee Retirement Income Security Act of 1974, as amended.
- (b) $\,\,$ Deal with the assets of the Trust in his own interests or for his own account.
- (c) Take part in any transaction, either in their individual or other capacities, involving the Trust on behalf of a party (or represent a party) whose interests are adverse to the interests of the Trust or the interests of its Participants or their beneficiaries.
- (d) Receive any consideration for his own personal accounts from any party dealing with the Trust in connection with a transaction involving the assets of the Trust.

Notwithstanding the foregoing prohibited transactions, any transaction which is not prohibited under applicable laws or for which an exemption has been obtained in accordance with the provisions of such applicable laws, may be engaged in by the Fiduciaries provided such transaction is for the exclusive benefit of the Participants and beneficiaries.

12.5 The Trustees shall keep accurate accounts of all investments, receipts and disbursements and other transactions hereunder, and all accounts, books and records relating thereto shall be open at all reasonable times to inspection and audit by any person or persons designated by the Committee and/or the Board of Directors.

-27-

28

CONFORMED COPY

- 12.6 (a) Within ninety (90) days following the close of each Plan Year, the Trustees and/or Plan Fiduciary employed by the company shall file with the committee a written account setting forth all securities and other property purchased and sold, all receipts, disbursements and other transactions effected by them during such annual period, and showing the securities and other property held at the end of such period and their fair market value on that date. Within ninety (90) days from the date of filing such account, the committee may file with the Trustees and/or Plan Fiduciary employed by the company either its written approval or its written disapproval with the reasons therefore, of the account so rendered.
- (b) Upon the filing of such approval of the Trustees' account, or at the expiration of ninety (90) days after the filing of such account, if written disapproval shall not have been filed with the Trustees or Plan Fiduciaries employed by the company, the Trustees or Plan Fiduciaries employed by the company shall be relieved from all liability, responsibility and accountability to the Company and the Union, but only to the Company and the Union, for their acts as set forth in said account. The foregoing provision, however, shall not preclude the Trustees and/or Plan Fiduciaries employed by the company from having their accounts settled by a court of competent jurisdiction.

The Trust hereby created is part of a Plan for the exclusive benefit of the Participants, retired Participants and their beneficiaries, which Plan the Company intends shall qualify under Sections 401(a), 401(k) and 401(m) of the Internal Revenue Code, as amended, and, until advised to the contrary, the Trustees may assume that the Plan so qualifies and that the Trust is exempt from tax under Section 501 (a) of the Internal Revenue Code, as amended. However, any taxes that may be assessed on or in respect of the Fund shall be a charge against the Fund. The Trustees may assume that any taxes assessed on or in respect of the Fund are lawfully assessed unless the committee shall in writing advise the Trustees that in the opinion of its counsel, such taxes are not lawfully assessed. In the event that the Committee shall so advise the Trustees, the Trustees, if so required by the Committee, shall contest the validity of such taxes in any manner deemed appropriate by the Committee or its counsel, but at the expense of the Company; or the Company may itself contest the validity of any such taxes in any manner deemed appropriate by it or its counsel. The word "taxes" in this paragraph shall be deemed to include any interest or penalties that may be levied or imposed in respect to any taxes.

ARTICLE XIII QUALIFICATION, AMENDMENT AND TERMINATION

- 13.1 It is the intention of the Company that the Plan and Trust shall be and remain qualified under Sections 401(a), 401(k), and 401(m) of the Internal Revenue Code, as amended. The Board of Directors may authorize any modification or amendments of the Plan or Trust, which may be retroactive, deemed necessary or appropriate in its opinion to qualify or maintain the Plan and Trust as a plan and trust meeting the requirements of Sections 401(a), 401(k), and 401(m) of the Internal Revenue code, as amended, or any other applicable provisions of the Internal Revenue code, as now in effect or hereafter amended or adopted and the regulations issued thereunder.
- 13.2 The company reserves the right at any time to modify, suspend, amend or terminate the Plan and/or Trust in whole or in part (including the provisions relating to contributions) by delivering to the committee and the Trustee a copy of such modification, suspension, amendment or termination as approved by the Board of Directors; provided that the company shall have no power to modify, suspend, amend or terminate the Plan or the Trust in such manner as will cause or permit any part of the Trust Fund to be used for or diverted to purposes other than for the exclusive benefit of Participants or their beneficiaries, except as otherwise provided in Section 6.4(b), or the payment of expenses pursuant to the provisions of the Plan, or as will cause or permit any portion of the Fund to revert to or become the property of the Company. notwithstanding anything herein contained, the company, upon such termination of the Plan, shall have no obligation or liability whatsoever to make any further payments to the Trustees, and neither the Trustees, nor any Participant, Employee or other person shall have any right to compel the Company to make any further payments.
- 13.3 In the event that contributions to the Plan are suspended or the Plan is terminated as hereinabove provided

-28-

29

CONFORMED COPY

and unless the Company shall also terminate the Trust, the Trust shall continue in existence, and the Fund shall be held, administered and distributed by the Trustee as directed by the committee as provided in this Agreement, and all of the provisions of the Plan and Trust set forth in this Agreement, which are applicable in the opinion of the committee and the Trustee, other than the provisions relating to contributions, shall remain in full force and effect. Upon complete discontinuance of contributions under the Plan, termination or partial termination of the Plan, the rights of all affected Participants to their Account Balances shall be nonforfeitable. In the case of a partial termination, nonforfeitable rights shall only be applicable to the portion of the Plan that is terminated.

A partial termination shall occur when a significant percentage of Participants have terminated employment in any Plan Year. A significant percentage shall be construed utilizing the principles set forth in Revenue Rulings 72-439, 72-510, 73-284, and 81-27.

- 13.4 In the event that both the Plan and Trust are terminated, at the direction of the Committee, the Trustees shall distribute all assets remaining in the Fund, after payment of the expenses properly chargeable against the Fund, to the Participants, or their beneficiaries, in accordance with the value of the accounts of such Participants as of the date of distribution, in cash or company stock or partly in cash and partly in Company stock and in such manner as the committee and the Trustee shall determine. The Committee's determination shall be final and conclusive on all persons. In determining the value of the accounts of the Participants as of the date of distribution, the assets of the Fund shall be valued by the Trustee at fair market value as of the close of business on the distribution date, and the accounts of the Participants shall be adjusted, in the manner provided in Article VI.
- 13.5 In the event that the Plan and Trust are terminated, the committee and the Trustee serving at the time of such termination shall continue to act with their full power hereunder until the completion of the distribution of such assets; and if the company and Union have been dissolved, a majority of the committee then serving shall have the power to fill any vacancies occurring among them after such termination by resignation, death or otherwise.
- 13.6 If the Plan shall be merged or consolidated with, or transfer in whole or in part assets or liabilities to any other plan, then each Participant in this Plan shall be entitled to receive a benefit immediately after the merger, consolidation or transfer, which is equal to or greater than the benefit he would have been entitled to receive immediately before the merger, consolidation or transfer, if this Plan had then terminated.

ARTICLE XIV MISCELLANEOUS

- 14.1 This Plan is based upon the condition precedent that it shall be approved and qualified under the Internal Revenue code and regulations issued thereunder with respect to employees' trusts, so as to permit the Company to deduct for Federal income tax purposes, the amount of its contributions to the Fund. If this Plan is not so initially approved and qualified with such amendments as the Internal Revenue Service may request and the company may consent to, then, notwithstanding any provisions to the contrary, the Trustees shall return to the Company within one (I) year any unused contributions, funds or assets held by the Trustees under this Plan. The return of the Company's contributions shall be conditioned upon the determination letter request being made to the Internal Revenue Service on a timely basis.
- 14.2 Neither the establishment of the Trust created hereby, nor any modification thereof, nor the creation of any fund or account, nor the payment of any benefits, shall be construed as giving to any Participant or other person any legal or equitable right against the Company, or any officer or Employee thereof, or the Trustees, except as herein expressly provided. All contributions made by the Company to the Plan, except Company contributions as a result of Employee Salary Deferral contributions and Voluntary contribution, shall be voluntary, and the Company shall be under no legal liability to make any such contributions. Nothing herein contained shall entitle any person to any payment except out of the Fund.
- 14.3 Neither the establishment of the Plan and Trust nor the granting of benefits nor any action of the Board of

Directors, the Committee or any Trustee now or hereafter, shall be held or construed to confer upon any person any legal right to be continued as an Employee, or to interfere with the right of the company to discharge any Employee, whenever the interests of the company in its sole discretion may so require, without liability to the company or any Trustee.

- The right of any Participant or beneficiary to any benefits or to any payment hereunder or to any separate account will not be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, charge or seizure. This does not preclude any Trustee at the direction of the Committee from complying with a court order requiring deduction from the benefits of a Participant in pay status for alimony or support payments pursuant to a qualified domestic relations order as defined in Section 414(p) of the Internal Revenue Code. If such Participant or beneficiary shall attempt to assign, transfer, or otherwise dispose of or encumber such right, or should such right become subject to attachment, execution, garnishment, sequestration, or other legal, equitable or other process, it shall ipso facts pass to each one or more persons as may be selected by the committee from among the beneficiaries, if any, theretofore designated by such Participant, or the spouse, blood relatives and adopted children of the Participant; PROVIDED, HOWEVER, that the committee in their sole discretion may reappoint the Participant or beneficiary to receive any payment thereafter becoming due either in whole or in part. Any appointment made by the Committee hereunder may be revoked by the Committee at any time, and a further appointment made by them. Payments made pursuant to this Section shall be made for the benefit and on behalf of the Participant and shall completely discharge the Company, the Union, the Committee, the Plan Administrator and the Trustees.
- 14.5 The facts as shown by the records of the Committee at the time of death of any person entitled to any benefits or to any payment hereunder shall be conclusive as to the identity of the proper payee and of the amounts properly payable, and payment in accordance with such state of facts shall constitute a complete discharge of any and all obligations under Article VIII hereof.
- 14.6 Whenever the Company under the terms of this Plan is permitted or required to do or perform any act, it shall be done and performed by any officer authorized by the Board of Directors.
- 14.7 The provisions of this Plan and Trust, and the rights and obligations herein created, shall be interpreted in accordance with the substantive law of the State in which the company was incorporated, except where such law is superseded by the Employee Retirement Income Security Act of 1974, as amended.
- 14.8 Notwithstanding anything herein to the contrary, upon the Company's request and upon notification of the Union, a contribution which was made under a mistake of fact or based upon the deductibility of the contribution under Section 404 of the Internal Revenue Code, as amended, shall be returned to the company within one (1) year after the payment of the contribution or the disallowance of the deduction (to the extent disallowed), whichever is applicable. The amount to be returned to the Company shall be equal to the actual contribution less the proportionate share of the losses to the Trust Fund attributable to such contribution.
- 14.9 Notwithstanding Section 14.4 of the Plan, the committee may direct the Trustees to comply with a qualified domestic relations order as defined in Section 414(p) of the Internal Revenue Code and other domestic relations orders permitted by the Plan Administrator under the provisions of the Retirement Equity Act of 1984. After the Committee has determined the qualified status of the domestic relations order, they shall administer and proceed with Plan distributions in accordance with the order pursuant to Article VIII of the Plan.

For purposes of this Plan, a qualified domestic relations order shall mean any judgment, decree, or order (including approval of a property settlement agreement) which has been determined by the Plan Administrator in accordance with the procedures established under the Plan to constitute a qualified domestic relations order within the meaning of Section 414(p) (1) of the Internal Revenue Code. All payments made pursuant to this Section in accordance with a qualified domestic relations order shall be made for the benefit and on behalf of the Participants and/or the alternate payee and shall completely discharge the company, the committee, the Plan Administrator and the

-30-

31

CONFORMED COPY

ARTICLE XV ROLLOVERS OR TRANSFERS

[SECOND AMD REVISED SECTION 15.1 IN ITS ENTIRETY, EFFECTIVE 1/1/93]

15.1 If permitted by the Committee, any Employee who upon meeting the Eligiblity provisions of Section 4.1 of the Plan would automatically become a Participant in the Plan, or any current Participant, upon approval of the Committee, may make a rollover or transfer to the Trustees of assets held under his name from a plan qualified under Section 401 of the Internal Revenue Code.

[SECOND AMD REVISED THE FIRST SENTENCE OF SECTION 15.2 TO READ AS FOLLOWS, EFFECTIVE 1/1/93]

- 15.2 Upon receipt of written approval from the Committee, the Trustee may accept a rollover or transfer of a Participant's interest from a qualified plan to a Rollover Account in this Plan. Such rollover or transfer shall represent part or all of the assets held for his benefit under a plan qualified under Section 401(a) of the Internal Revenue Code. Under no circumstances may amounts required to be distributed to a Participant under the minimum distribution Regulations of Section 401(a) (9) of the Internal Revenue Code be rolled over or transferred to this Plan.
- 15.3 The Participant shall notify the Committee that the payment is a rollover or direct plan to plan transfer and provide such other statements and information as may be required by the Committee in order to establish that such rollover or transfer meets the requirements of the Internal Revenue Code and the Regulations promulgated thereunder.
- $15.4\,$ The Trustee shall credit such rollover or transfer to a Rollover Account maintained hereunder for such Participant.
- 15.5 A Participant's Rollover Account shall at all times be one hundred (100%) percent vested and nonforfeitable. Such account shall be valued and adjusted as hereinafter set forth. In the event of a Participant's termination of employment for any reason whatsoever, or in the event of termination of the Plan, a Participant's Rollover Account balance shall be deemed to be part of the Participant's Account Balance and shall be distributed to him or his beneficiary, as the case may be, in accordance with the provisions contained in Article VIII.
- 15.6 The assets credited to the Rollover Account shall be segregated by the Trustees. The Trustees shall have the same rights and powers regarding such segregated assets as they have with respect to the Trust Fund. The Trustees shall value such assets at their fair market value as of the Valuation Date. The Trustees' determination of the value of the assets shall be final and conclusive for all purposes of the Plan.
- 15.7 The amount which shall be considered credited to the Rollover Account of each Participant on each Valuation Date shall be:
- $\hbox{ (a)} \qquad \hbox{The Rollover Account balance on the prior Valuation} \\ \hbox{Date; plus}$
- (b) Any rollovers or transfers on behalf of the Participant made in the Plan Year immediately prior to the current Valuation Date; plus or minus
- (c) The increase or decrease since the preceding Valuation Date in the value of the assets credited to his Rollover Account.

15.8 The Rollover Account balance of any Participant whose employment shall terminate for whatever reason shall be determined as of the date of termination of employment as if such date were a Valuation Date, in accordance with the provisions of this Article.

-31-

32

CONFORMED COPY

ARTICLE XVI PARTICIPANT'S INVESTMENT ELECTIONS

- 16.1 (a) If so permitted by the Company and the Committee, a Participant may notify the Committee, in writing that the manner in which his Employee Salary Deferral Contributions, Voluntary Contributions and Rollover Account shall be invested by the Trustee.
- (b) All Company contributions shall be deposited by the Trustee in THE DURIRON STOCK FUND. Each Participant's Employee Salary Deferral Contributions and Voluntary Contributions shall be deposited by the Trustee in the Equity Fund and/or in the Fixed Income Fund, as the Committee shall direct; PROVIDED, HOWEVER, that the Committee's directions shall be based upon the investment election of each Participant, made in accordance with the provisions of Section 16.2.
- 16.2 A Participant's investment election shall specify his choice among the investment funds selected by the Committee as long as the allocation of the Participant's election is in ten (10) percentage point denominations. Notwithstanding anything the contrary, if the Committee selects other funds for the investment of the Participants' contributions to the Plan, the Committee shall distribute an administrative communication which shall supersede the provisions and limitations of this Section 16.2.
- 16.3 The investment option so elected by a Participant shall remain in effect until he ceases to be a Participant in accordance with the provisions of the Plan; PROVIDED, HOWEVER, in accordance with the procedures and limitations of Sections 4.6 and 4.7, that a Participant may change his investment election as of any Withholding Change Date by filing with the Committee, a written election directing a change in his investment election. Any such change shall not affect the amounts credited to any Account of such Participant as of the day immediately preceding the Withholding Change Date as of which such change is to become effective except with respect to any Employee Salary Deferral and Voluntary Contributions made thereafter.
- 16.4 The Trustee shall establish a minimum three (3) funds hereunder, the Equity Fund, the Fixed Income Fund, and THE DURIRON STOCK FUND. At the discretion of the Committee, additional funds may be established for the investment of a Participant's Employer Salary Deferral Contributions, Voluntary Contributions, and Rollover Account contributions to the Plan. Such Funds shall be invested in accordance with the investment guidelines set forth in Sections 4.6, 4.7 and 16.3. All assets of the Trust, shall be held and administered by the Trustee through the media of such Funds as separate common trust funds of the Trust. The interest of each Participant, former Participant, or beneficiary under the Plan in any of such Funds shall be an undivided interest.
- 16.5 Any dividends, interest, distributions, or other income received by the Trustee in respect of either the Equity Fund, the Fixed Income Fund, or THE DURIRON STOCK FUND, shall be reinvested by the Trustee in the Fund in respect of which such income was received by it.
- 16.6 Each Participant will have the right to direct the Trustee to vote the stock (i.e., shares of stock issued by the Company or any related entity as defined by Section 414 of the Internal Revenue Code) in his Employer Contribution Account on all matters requiring a vote. The Committee will cause

such Participants to receive any materials supplied to shareholders in connection with any such vote and will do all other things necessary to enable such Participants to vote such Stock. The Trustee shall vote any such shares of Stock for which it receives no directions and any shares of stock that are not allocated to Participant Employer Contribution Accounts, in proportion to the directions received with respect to shares which are allocated to Participant Employer Contribution Accounts and for which it has received directions.

In addition, if the Trustee receives an offer to acquire the stock held in the Trust Fund, each Participant will have the right to direct the Trustee to sell or to refrain from selling all or any portion of the stock in his Employer Contribution Account. The Committee will use its best efforts to cause such Participants to receive all materials supplied to shareholders in connection with such offer to acquire stock. The Participant will direct the Trustee as to what percentage, if any, of the total stock in his

-32-

33

CONFORMED COPY

Employer Contribution Account is to be sold and, at what price it is to be sold. The Trustee will exercise its discretion concerning the sale of stock for which it receives no directions or any shares of stock not allocated to participants' Employer Contribution Accounts. If the Trustee, in its discretion determines that it is in the Participants' best interests to sell, the Trustee will sell the shares of stock at a price that is consistent with what the Trustee deems to be in the best interests of the Participants.

Any proceeds received from the sale of stock will be used to purchase additional stock at the then current market price.

- 16.7 In the event of the death of a Participant, all rights of management shall thereupon revert to the Committee.
- 16.8 The Committee and the Trustee shall incur no liability whatsoever for any action of a Participant taken pursuant to this Article and shall be indemnified and held harmless by such Participant against any liability for any act or failure to act in any manner pursuant to the direction of such Participant.

[THIRD AMENDMENT ADDED ARTICLE XVII, EFFECTIVE AS OF JANUARY 1, 1994]

ARTICLE XVII LOANS TO PARTICIPANTS

- 17.1 The Committee, in its sole discretion and upon written request by a Participant, may, if it finds him to be creditworthy, direct the Trustee to make a loan (of an amount that is not less than \$1,000) to him from the Trust Fund for any purpose which the Committee has designated in its written loan policy. The amount of any loan, together with the unpaid balances of any other outstanding loans from the Plan and accrued interest thereon, shall not exceed an amount equal to the lesser of:
- (a) \$50,000 (reduced by an amount equal to any loan balance repaid to the Plan during the immediately preceding 12-month period); or
- (b) 50% of the Participant's Account Balance (less that portion of the Account Balance which is invested in the Duriron Stock Fund, as described in Article XVI), determined as of the last preceding Valuation Date.
- 17.2 Each such loan shall be governed by the written loan policies adopted by the Committee and shall be evidenced by a written note containing such terms as may be determined by the Trustee. Such terms shall include provision for payment of interest, shall include a provision for payment within 5 years after the loan was granted and shall require substantially level amortization of the loan (with payments by payroll deduction, but not less frequently than quarterly) over the period of the loan.

The Committee shall establish a Loan Account in the name of each Participant to whom a loan is made, which Loan Account shall be periodically adjusted by charging to the Participant's Loan Account any principal and interest received by the Trustee on account of the Participant's loan since the last preceding Valuation Date.

17.3 If a loan or portion of a loan made to the Participant, together with the accrued interest thereon, remains unpaid at such time as the Participant receives a distribution from the Plan, an amount equal to such loan or any part thereof, together with the accrued interest thereon, shall be charged to the Participant's Loan Account as of the Valuation Date coincident with or next preceding such distribution, after all other adjustments required under the Plan have been made.

-33-

34

CONFORMED COPY

[THIRD AMENDMENT ADDED ADDENDUM A]

ADDENDUM A

IRS MODEL SECTION 401(a)17 LIMITATION

In addition to other applicable limitations set forth in the plan, and notwithstanding any other provision of the plan to the contrary, for plan years beginning on or after January 1, 1994, the annual compensation of each employee taken into account under the plan shall not exceed the OBRA '93 annual compensation limit. The OBRA '93 annual compensation limit is \$150,000, as adjusted by the Commissioner for increases in the cost of living in accordance with section 401(a)(17)(B) of the Internal Revenue Code. The cost-of-living adjustment in effect for a calendar year applies to any period, not exceeding 12 months, over which compensation is determined (determination period) beginning in such calendar year. If a determination period consists of fewer than 12 months, the OBRA '93 annual compensation limit will be multiplied by a fraction, the numerator of which is the number of months in the determination period, and the denominator of which is 12.

For plan years beginning on or after January 1, 1994, any reference in this plan to the limitation under section 401(a)(17) of the Code shall mean the OBRA '93 annual compensation limit set forth in this provision.

If compensation for any prior determination period is taken into account in determining an employee's benefit accruing in the current plan year, the compensation for that prior determination period is subject to the OBRA '93 annual compensation limit in effect for that prior determination period. For this purpose, for determination periods beginning before the first day of the first plan year beginning on or after January 1, 1994, the OBRA '93 annual compensation limit is \$150,000.

A-1

35

CONFORMED COPY

[THIRD AMENDMENT ADDED ADDENDUM B]

ADDENDUM B

IRS MODEL NOTIFICATION REQUIREMENTS

If a distribution is one to which sections 401(a)(11) and 417 of the Internal

Revenue Code do not apply, such distribution may commence less than 30 days after the notice required under section 1.411(a)-11(c) of the Income Tax Regulations is given, provided that:

- (1) the plan administrator clearly informs the participant that the participant has a right to a period of at least 30 days after receiving the notice to consider the decision of whether or not to elect a distribution (and, if applicable, a particular distribution option), and
- (2) the participant, after receiving the notice, affirmatively elects a distribution.

[FLOWSERVE LETTERHEAD]

June 22, 1998

Flowserve Corporation 222 W. Las Colinas Boulevard Suite 1500 Irving, Texas 75039

Dear Sirs:

With reference to the registration statement on Form S-8 which Flowserve Corporation (the "Company") proposes to file with the Securities and Exchange Commission (the "SEC") under the Securities Act of 1933, registering 50,000 shares of common stock, par value \$1.25 per share, of the Company (the "Shares") which may be offered and sold by the Company under the Duriron Company, Inc., Pump and Foundry Divisions Hourly Employees Savings and Thrift Plan (collectively, the "Plan"), I am of the opinion that:

- the Company is a corporation duly organized, validly existing and in good standing under the laws of the State of New York, and
- 2. all proper corporate proceedings have been taken so that any Shares to be offered and sold which are of original issuance, upon sale and payment therefor in accordance with the Plan and the resolutions of the Board of Directors relating to the offering and sale of the Shares thereunder, will be legally issued, fully paid and nonassessable.

I hereby consent to the filing of this opinion with the SEC in connection with the registration statement referred to above.

Very truly yours,

/s/ RONALD F. SHUFF

Ronald F. Shuff Vice President, Secretary & General Counsel

CONSENT OF INDEPENDENT AUDITORS

We consent to the incorporation by reference in Flowserve Corporation's Registration Statement on Form S-8 pertaining to The Duriron Company, Inc., Pump and Foundry Divisions Hourly Savings and Thrift Plan of our reports dated February 20, 1998, with respect to the consolidated financial statements of Flowserve Corporation incorporated by reference in its Annual Report (Form 10-K) for the year ended December 31, 1997 and the related financial statement schedule included therein, filed with the Securities and Exchange Commission.

/s/ Ernst & Young LLP

Dallas, Texas June 22, 1998

EXHIBIT 23.3

CONSENT OF INDEPENDENT ACCOUNTANTS

We hereby consent to the incorporation by reference in this Registration Statement on Form S-8 of our report (relating to BW/IP, Inc. and its subsidiaries) dated January 28, 1997 appearing on page F-2 of Flowserve Corporation's Annual Report on Form 10-K for the year ended December 31, 1997.

/s/ PRICE WATERHOUSE LLP

Los Angeles, California June 22, 1998

FLOWSERVE CORPORATION

POWER OF ATTORNEY

The undersigned does hereby make, constitute and appoint Ronald F. Shuff with full power to act as her true and lawful attorney, in her name, place and stead to execute on her behalf, as a director of Flowserve Corporation (the "Company"), the Company's Registration Statement on Form S-8 for shares of the Company's common stock to be issued pursuant to The Duriron Company, Inc., Pump and Foundry Divisions Hourly Employees Savings and Thrift Plan, and any and all amendments thereto, to be filed with the Securities and Exchange Commission (the "SEC") pursuant to the provisions of the Securities Act of 1933, as amended, and the rules and regulations of the SEC promulgated thereunder, together with any other instruments that such attorney shall deem necessary or advisable in connection therewith, giving and granting to such attorney full power and authority to do and to perform every act necessary or advisable in furtherance of the purposes hereof as fully as she could do herself, with full power of substitution and revocation, hereby ratifying and confirming all that such attorney may or shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has hereunto set her hand on the date indicated below.

/s/ Diane C. Harris
----Diane C. Harris

Dated: June 18, 1998

2

FLOWSERVE CORPORATION

POWER OF ATTORNEY

The undersigned does hereby make, constitute and appoint Ronald F. Shuff with full power to act as his true and lawful attorney, in his name, place and stead to execute on his behalf, as a director of Flowserve Corporation (the "Company"), the Company's Registration Statement on Form S-8 for shares of the Company's common stock to be issued pursuant to The Duriron Company, Inc., Pump and Foundry Divisions Hourly Employees Savings and Thrift Plan, and any and all amendments thereto, to be filed with the Securities and Exchange Commission (the "SEC") pursuant to the provisions of the Securities Act of 1933, as amended, and the rules and regulations of the SEC promulgated thereunder, together with any other instruments that such attorney shall deem necessary or advisable in connection therewith, giving and granting to such attorney full power and authority to do and to perform every act necessary or advisable in furtherance of the purposes hereof as fully as he could do himself, with full power of substitution and revocation, hereby ratifying and confirming all that such attorney may or shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has hereunto set his hand on the date indicated below.

/s/ George T. Haymaker, Jr.
George T. Haymaker, Jr.

Dated: June 18, 1998

3

FLOWSERVE CORPORATION

POWER OF ATTORNEY

The undersigned does hereby make, constitute and appoint Ronald F. Shuff with full power to act as his true and lawful attorney, in his name, place and stead to execute on his behalf, as a director of Flowserve Corporation (the "Company"), the Company's Registration Statement on Form S-8 for shares of the Company's common stock to be issued pursuant to The Duriron Company, Inc., Pump and Foundry Divisions Hourly Employees Savings and Thrift Plan, and any and all amendments thereto, to be filed with the Securities and Exchange Commission (the "SEC") pursuant to the provisions of the Securities Act of 1933, as amended, and the rules and regulations of the SEC promulgated thereunder, together with any other instruments that such attorney shall deem necessary or advisable in connection therewith, giving and granting to such attorney full power and authority to do and to perform every act necessary or advisable in furtherance of the purposes hereof as fully as he could do himself, with full power of substitution and revocation, hereby ratifying and confirming all that such attorney may or shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has hereunto set his hand on the date indicated below.

Dated: June 18, 1998

4

FLOWSERVE CORPORATION

POWER OF ATTORNEY

The undersigned does hereby make, constitute and appoint Ronald F. Shuff with full power to act as his true and lawful attorney, in his name, place and stead to execute on his behalf, as a director of Flowserve Corporation (the "Company"), the Company's Registration Statement on Form S-8 for shares of the Company's common stock to be issued pursuant to The Duriron Company, Inc., Pump and Foundry Divisions Hourly Employees Savings and Thrift Plan, and any and all amendments thereto, to be filed with the Securities and Exchange Commission (the "SEC") pursuant to the provisions of the Securities Act of 1933, as amended, and the rules and regulations of the SEC promulgated thereunder, together with any other instruments that such attorney shall deem necessary or advisable in connection therewith, giving and granting to such attorney full power and authority to do and to perform every act necessary or advisable in furtherance of the purposes hereof as fully as he could do himself, with full power of substitution and revocation, hereby ratifying and confirming all that such attorney may or shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has hereunto set his hand on the date indicated below.

/s/ William C. Rusnack

Dated: June 18, 1998

5

FLOWSERVE CORPORATION

POWER OF ATTORNEY

The undersigned does hereby make, constitute and appoint Ronald F. Shuff with full power to act as his true and lawful attorney, in his name, place and stead to execute on his behalf, as a director of Flowserve Corporation (the "Company"), the Company's Registration Statement on Form S-8 for shares of the Company's common stock to be issued pursuant to The Duriron Company, Inc., Pump and Foundry Divisions Hourly Employees Savings and Thrift Plan, and any and all amendments thereto, to be filed with the Securities and Exchange Commission (the "SEC") pursuant to the provisions of the Securities Act of 1933, as amended, and the rules and regulations of the SEC promulgated thereunder, together with any other instruments that such attorney shall deem necessary or advisable in connection therewith, giving and granting to such attorney full power and authority to do and to perform every act necessary or advisable in furtherance of the purposes hereof as fully as he could do himself, with full power of substitution and revocation, hereby ratifying and confirming all that such attorney may or shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has hereunto set his hand on the date indicated below.

/s/ Kevin E. Sheehan

Kevin E. Sheehan

Dated: June 18, 1998

6

FLOWSERVE CORPORATION

POWER OF ATTORNEY

The undersigned does hereby make, constitute and appoint Ronald F. Shuff with full power to act as his true and lawful attorney, in his name, place and stead to execute on his behalf, as a director of Flowserve Corporation (the "Company"), the Company's Registration Statement on Form S-8 for shares of the Company's common stock to be issued pursuant to The Duriron Company, Inc., Pump and Foundry Divisions Hourly Employees Savings and Thrift Plan, and any and all amendments thereto, to be filed with the Securities and Exchange Commission (the "SEC") pursuant to the provisions of the Securities Act of 1933, as amended, and the rules and regulations of the SEC promulgated thereunder, together with any other instruments that such attorney shall deem necessary or advisable in connection therewith, giving and granting to such attorney full power and authority to do and to perform every act necessary or advisable in furtherance of the purposes hereof as fully as he could do himself, with full power of substitution and revocation, hereby ratifying and confirming all that such attorney may or shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has hereunto set his hand on the date indicated below.

Dated: June 18, 1998

7

FLOWSERVE CORPORATION

POWER OF ATTORNEY

The undersigned does hereby make, constitute and appoint Ronald F. Shuff with full power to act as his true and lawful attorney, in his name, place and stead to execute on his behalf, as a director of Flowserve Corporation (the "Company"), the Company's Registration Statement on Form S-8 for shares of the Company's common stock to be issued pursuant to The Duriron Company, Inc., Pump and Foundry Divisions Hourly Employees Savings and Thrift Plan, and any and all amendments thereto, to be filed with the Securities and Exchange Commission (the "SEC") pursuant to the provisions of the Securities Act of 1933, as amended, and the rules and regulations of the SEC promulgated thereunder, together with any other instruments that such attorney shall deem necessary or advisable in connection therewith, giving and granting to such attorney full power and authority to do and to perform every act necessary or advisable in furtherance of the purposes hereof as fully as he could do himself, with full power of substitution and revocation, hereby ratifying and confirming all that such attorney may or shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has hereunto set his hand on the date indicated below.

Dated: June 18, 1998